Advancements in Green Hydrogen Production: A Comprehensive Review of System Integration, Power Grid Applications, and Cost Optimization

Hossam Ashraf1*, Masahiro Mae1, and Ryuji Matsuhashi1

¹Department of Electrical Engineering and Information Systems, The University of Tokyo, Tokyo 113-8656, Japan.

*Corresponding author

Email: hossam-ashraf@g.ecc.u-tokyo.ac.jp, mmae@g.ecc.u-tokyo.ac.jp, matu@g.ecc.u-tokyo.ac.jp

Abstract- Hydrogen is acquiring a promising recognition as a new trend in energy storage technologies due to its advantageous features including fast response, high energy density, and unconstrained storage capacity. Thus, it offers an effective solution for addressing the stability challenges posed by the large-scale integration of renewable energy sources (RESs) into power systems. Accordingly, this paper presents a comprehensive review of advancements in green hydrogen production (GHP), with a focus on water electrolyzers (WELs) and their integration into power systems. Specifically, it examines WEL types, operational characteristics, and the role of DC converters in system connectivity and efficient power flow. Furthermore, various control strategies to optimize converter performance are thoroughly analysed, along with WEL applications in frequency and voltage regulation, congestion management, and black start operations. Moreover, recent efforts to minimize hydrogen production costs through optimal system configurations and resource management are reviewed. It's worth indicating that alkaline WELs have the lowest capital cost, qualifying them as a cost-effective option for large-scale hydrogen production. Therefore, this article seeks to aid researchers and stakeholders by providing an insightful overview of the present status of WEL emergence in modern energy systems, highlighting key technological advancements, challenges, and prospects.

Keywords: Renewable energy sources; Green hydrogen production; Water electrolyzer; DC converters; Power system stability.

List of Abbreviations

AEM	:	Anion exchange membrane	LCOH	:	Levelized cost of hydrogen
AL	:	Alkaline	Li-ion	:	Lithium ion
ANFIS	:	Adaptive neuro-fuzzy inference system	LTTS	:	Low temperature thermal storage
BBC	:	Buck-boost converter	MPC	:	Model predictive control
CAES	:	Compressed air energy storage	Na-s	:	Sodium-sulfur
CES	:	Cryogenic energy storage	Ni-cd	:	Nickel-cadmium
DC converter	:	DC-to-DC converter	NPV	:	Net present value
DER	:	Distributed energy generator	PEM	:	Proton exchange membrane
DL	:	Deep learning	PHSS	:	Pumped hydro storage system
DTR	:	Dynamic thermal rating	PID	:	Proportional integral derivative
EST	:	Energy storage technology	PV	:	Photovoltaic

EV	:	Electric vehicle	RES	:	Renewable energy system
FC	:	Fuel cell	RL	:	Reinforcement learning
FESS	:	Flywheel energy storage system	SESS	:	Supercapacitor energy storage system
FIBBC	:	Floating-interleaved buck-boost converter	SL	:	Supervised learning
FLC	:	Fuzzy logic control	SO	:	Solid oxide
FNN	:	Feedforward neural network	SMC	:	Sliding mode control
GHP	:	Green hydrogen production	SMSS	:	Superconducting magnetic storage system
HESS	:	Hydrogen energy storage system	STSMC	:	Super-twisting sliding mode control
HOSMC	:	Higher-order sliding mode control	TSMC	:	Terminal sliding mode control
HTTS	:	High temperature thermal storage	UL	:	Unsupervised learning
IBBC	:	Interleaved buck-boost converter	VR	:	Vanadium redox
IRR	:	Internal rate of return	WEL	:	Water electrolyzer
LA	:	Lead acid	WT	:	Wind turbine
LCOE	:	Levelized cost of electricity	Zn-Br	:	zinc-bromine

1. Introduction

Fossil fuel consumption has raised significant global concerns due to severe ecological drawbacks and greenhouse gas emissions that accelerate climate change [1]. These fuels also contribute to water pollution and habitat destruction, and their unsustainability raises doubts about long-term availability and price stability [2]. Consequently, the global energy landscape is shifting toward renewable energy sources (RESs), such as photovoltaic (PV), solar thermal, wind turbine (WT), hydro, geothermal, and biomass, to reduce carbon emissions and achieve energy security [3-5]. Practically, governments and agencies worldwide have implemented regulations such as the Paris Agreement to boost RES adoption [6]. For instance, Japan has committed to carbon neutrality by 2050, marking a significant step beyond earlier targets [7].

While RESs like PV and WT provide abundant clean energy, their dependence on fluctuating weather conditions can pose stability challenges for modern power grids [8]. Accordingly, energy storage technologies (ESTs) help mitigate these issues by storing excess power during light-load conditions and releasing it when demand increases [9]. These ESTs may employ electrochemical, thermal, electromechanical, or electromagnetic methods [10-12]. Among them, hydrogen energy storage systems (HESS) derived from green hydrogen production (GHP), where hydrogen is generated from RESs using water electrolyzers (WELs), present a particularly promising solution for carbon emission reduction and power system stabilization [13]. Specifically, hydrogen offers high energy density, can be stored in gaseous or liquefied form, and is non-toxic, thus it enables flexible use across multiple applications [14, 15]. Moreover, it can be reconverted into electricity using fuel cells (FCs), signifying its role as a cornerstone of a carbon-neutral society [16].

In fact, incorporating WEL/FC-based systems within power grids can facilitate demand-supply balance by converting surplus energy into hydrogen, which is later used during peak demand or reduced renewable output [17, 18]. This dynamic flexibility aids frequency regulation, voltage control, and overall grid resilience [19]. However, efficient integration requires suitable DC-to-DC converter (DC converter) topologies and robust control methodologies to manage power flow. Non-isolated converters are often employed in smaller-scale applications, while isolated converters provide electrical isolation for high-power systems [20, 21]. Additionally, advanced control strategies, ranging from proportional–integral (PI) and fuzzy logic to neural networks, are critical for addressing nonlinear system dynamics and ensuring stable operation [22-25].

In this regard, several reviews have addressed individual aspects of GHP, such as technical principles, market potential, and challenges [26-28], the role of WELs in decarbonized power grids [29, 30], and WEL/FC-based configurations in AC/DC networks [31]. Other studies focus on FC/WEL control strategies [32], examine broader energy storage options [33], or perform techno-economic assessments of different RESs and WEL types [34-36]. Yet, a comprehensive discussion that integrates both technical and economic dimensions, covering WEL/FC designs, converter topologies, control schemes, and profit optimization for power grids, is still lacking.

To fill this gap, the present review provides a holistic evaluation of GHP systems, emphasizing their role in power regulation, grid stability, and economic performance. The key contributions can be summarized as follows:

- a) A systematic comparison of various ESTs, highlighting why hydrogen-based solutions stand out.
- b) An in-depth analysis of different WEL types, e.g., alkaline (ALWEL), proton-exchange membrane (PEMWEL), anion exchange membrane (AEMWEL), and solid oxide (SOWEL), and their bestfit applications.
- c) A review of mathematical modelling approaches for WELs, with detailed formulations, particularly for PEMWEL.
- d) An overview of DC converter topologies (isolated and non-isolated) and their control strategies for efficient power transfer with WELs and FCs.
- e) A survey of practical WEL applications in modern power systems, focusing on stability and ancillary services.
- f) Recent approaches to optimizing economic performance when integrating GHP and RESs.
- g) Key insights into future research directions and challenges in achieving carbon-neutral power systems.

The remainder of the article is organized as follows: Section 2 discusses the concept, advantages, and disadvantages of various ESTs. Section 3 introduces the internal construction and chemical reactions

of common WEL types. Section 4 reviews WEL modelling approaches and provides a detailed mathematical formulation for PEMWEL. Section 5 explores DC converter topologies for connecting fuel cells and WELs to the grid, while Section 6 highlights corresponding control strategies. Section 7 illustrates WEL applications in modern power systems, and Section 8 addresses recent efforts to optimize economic performance. Finally, Section 9 concludes with key findings and future perspectives. A graphical overview of the scope is shown in Fig. 1.



Fig. 1. Organization of paper scope

2. Overview of various energy storage technologies

Since the growing integration of RESs in power systems is characterized by a continuous substitution of synchronous generators with static inverters, significant instability issues have arisen due to power fluctuations [37, 38]. Such power electronic devices reduce the system's mechanical inertia making it vulnerable to instability attacks as a result of any supply-demand imbalance [39]. Actually, minimizing the mechanical inertia increases the system efficiency, however, it can vigorously trigger frequency oscillations leading to recurring tripping of under/over frequency relays. Thus, the operator may resort to shed the loads or automatic series outages may occur [40].

Indeed, ESTs offer an effective solution to overcome the aforementioned problems related to the high penetration of RESs in power networks [41]. Principally, an EST-based system stores surplus energy in a form different from its original one for a while then releases it upon a call to support the power balance [42]. Fig. 2 classifies various ESTs employed in power systems according to the type of stored energy [43]. Specifically, hydrogen-based energy storage system (HESS) represents chemical ESTs, and supercapacitor energy storage system (SESS) and superconducting magnetic storage system (SMSS) are examples of electrical ESTs. Besides, electrochemical ESTs refer to batteries, such as lithium-ion (Li-ion), lead-acid (LA), nickel-cadmium (Ni-Cd), sodium-sulfur (Na-S), zinc-bromide (Zn-Br), and vanadium redox (VR) batteries. Additionally, mechanical ESTs include compressed air energy storage (CAES), flywheel energy storage system (FESS), and pumped hydro storage system (PHSS), while thermal ESTs comprise cryogenic energy storage (CES), low temperature thermal storage (LTTS), and high temperature thermal storage (HTTS). Furthermore, the technical characteristics of such technologies are thoroughly described in Table 1, which are extracted from [44-47].

Particularly, HESS is a promising technology where hydrogen serves as the energy carrier. Due to its abundance and ecological benefits, hydrogen supports transportation, power generation, heating, and industry. However, its sustainability depends on the production process [48]. Hence, the authors in [49] introduced a quality-based methodology to classify diverse hydrogen production techniques with a color coding, as revealed in Fig. 3. Among the four methods, green and blue hydrogen gained a wide popularity for attaining carbon neutrality [50]. Explicitly, green hydrogen attracts research interest due to its reversibility in power systems [51]. In detail, Excess RES electricity is converted into hydrogen via WELs, while fuel cells regenerate electricity from stored hydrogen during peak demand [52].



Fig. 2. Classifications of ESTs

EST	Commercial status	Installed power (MW)	Efficiency (%)	Capital cost (\$/W)	Charging period	Discharging period	Time span, Cycles × 10 ³ (years)
HESS	Growing	0-58.8	25-58	0.5-10	hr - months	sec - days	1-20 (5-20)
SESS	Growing	0-0.3	90-95	0.1-0.45	sec - hr	msec - hr	>100 (20)
SMSS	Growing	0.1-10	95-98	0.2-0.489	min - hr	msec - 8 sec	>100 (20)
Li-ion	Commercialized	0-100	85-90	0.9-4	min - days	min - hr	1-20 (5-15)
LA	Mature	0-40	70-90	0.3-0.6	min - days	sec - hr	2 (3-15)
Ni-Cd	Commercialized	0-40	60-65	0.5-1.5	min - days	sec - hr	2-3.5 (10-20)
Na-S	Commercialized	0.05-34	80-90	1-3	sec - hr	sec - hr	2.5-4.5 (10-15)
Zn-Br	Demonstration	0.05-10	75 (average)	0.7-2.5	hr - months	sec - 10 hr	>2 (5-10)
VR	Pre-commercialized	0.03-3	85 (average)	0.6-1.5	hr - months	sec - 10 hr	>12 (5-10)
CAES	Mature	5-1000	70-89	0.4-1	hr - months	hr - days	>13 (20-40)
FESS	Pre-commercialized	0.1-20	93-95	0.25-0.35	sec - min	msec - 15 min	>100 (>15)
PHSS	Mature	100-5000	75-85	2-4.3	hr - months	hr - days	>13 (40-60)
CES	Growing	0.1-300	40-50	0.2-0.3	min - days	hr - 8 hr	>13 (20-40)
LTTS	Growing	0-5	50-90	-	min - days	hr - 8 hr	- (10-20)
HTTS	Growing	0-60	30-60	-	min - months	hr - days	>13 (5-15)

Table 1. Practical features of some ESTs

*sec, min, and hr refer to second, minute, and hour, respectively.



Fig. 3. Classification of hydrogen production systems

Besides, according to Table 1, HESS provides a longer-term energy solution than other ESTs, making it ideal for sustaining large loads during power outages. Its rapid response helps regulate power system frequency, mitigating RES intermittency effects. Additionally, modularity and scalability enhance its adaptability, while oxygen as a byproduct makes it an environmentally friendly option [53]. Recent studies suggest that integrating distributed RESs with HESS improves power reliability, resilience, and continuity. Table 2 highlights global HESS pilot projects, demonstrating its feasibility in power networks. However, high installation costs and the need for a robust hydrogen infrastructure remain key challenges [54].

			V	VEL	Hydrog	en tank	FC	
Ref.	Origin	RES' type	Туре	Capacity (kW)	Pressure (bar)	Volume (Nm ³)	Туре	Output power (kW)
[55]	Germany	PV	AL	100	30	5000	PAFC	80
[56]	Spain	PV	AL	5	10	24	PEMFC/PAFC	7.5/10
[57]	USA	PV	AL	6	5.7	30	×	×
[58]	Germany	PV	PEM	2	15	30	×	×
[59]	Germany	PV	AL	26	120	3000	PEMFC	5.6
[60]	Canada	PV/Wind	AL	5	120	40	PEMFC	5
[61]	Argentina	PV/Wind	AL	5	35	10	×	×
[62]	Spain	PV	PEM	1	70	30	×	×
[63]	Norway	Wind	PEM	1.5	14	16	×	×
[64]	France	PV	AL	3.6	0.4	10	×	×
[65]	Italy	PV	AL	3.4	4	10	×	×
[66]	USA	PV	PEM	3.35	15	5.4	PEMFC	2.4
[67]	Spain	PV	AL	15	25	2	×	×
[68]	Australia	×	PEM	100	163	14	×	×
[69]	UK	×	AL	250	NM	10	NM	100
[70]	UK	×	PEM	500	20	NM	×	×

 Table 2. Global HESS ongoing projects

"NM" and "x" refer to not mentioned and not existed in the reported article, respectively.

3. Advances in WEL technologies

A WEL utilizes a flow of electrons, injected from an external DC voltage source, to return water to its forming atoms, hydrogen (H₂) and oxygen (O₂) [71]. It basically comprises an electrolyte sandwiched by two electrodes (anode and cathode). So, passing an electrical current via water yields hydrogen at the cathode and oxygen at the anode [72]. Considering the electrolyte material, WELs have several types, such as ALWEL [73], AEMWEL [74], PEMWEL [75], and SOWEL [76]. To visualize the main differences of these types, Fig. 4 encloses their schematic diagrams. Technically, each type has distinctive characteristics, regarding operating temperature and pressure, efficiency, current density, input voltage, and lifetime, as indicated in Table 3 [77].



Fig. 4. Schematic diagram of WEL's types

Table 3. Practical features of various WEL's type	/pes
---	------

Type Feature	ALWEL	AEMWEL	PEMWEL	SOWEL
Electrolyte	KOH/NaOH	КОН	Solid polymer (Nafion)	yttria-stabilized zirconia (YSZ)
Carrier ion	OH-	OH-	H^{+}	O ²⁻
Temperature	65 – 100 °C	50 – 70 °C	70 − 90 °C	600 – 1000 °C
Pressure	2-10 bar	\leq 35 bar	15 – 30 bar	< 30 bar
Efficiency	62 - 84 %	50 - 70 %	67 - 82 %	$\leq 100 \%$
Voltage/cell	1.8 - 2.4 V	1.85 V	1.8 - 2.4 V	$0.95 - 1.30 \; V$
Current density	$0.2 - 0.4 \text{ A. cm}^{-2}$	$0.1 - 0.5 \text{ cm}^{-2}$	$0.6 - 2.0 \text{ cm}^{-2}$	$0.3 - 1.0 \text{ cm}^{-2}$
Response time	Seconds	Seconds	Milliseconds	Seconds
Lifetime	$< 90 \times 10^3$ hr	$> 10 \times 10^{3} \text{ hr}$	< 40 × 10 ³ hr	$<$ 40 \times 10 ³ hr
Scalability	High	Reasonable	Reasonable	Reasonable
Applicability status	Mature	Commercial	Commercial	Under development
		Electrochemical reactions		
At anode:	$40 H^- \to 0_2 + 2 H_2 0 + 4 e^-$	$40H^- \to 0_2 + 2H_20 + 4e^-$	$H_2 0 \rightarrow 2H^+ + \frac{1}{2}O_2 + 2e^-$	$0^{2-} \rightarrow \frac{1}{2}O_2 + 2e^-$
At cathode:	$2H_20+2e^- \rightarrow H_2+20H^-$	$2H_20+2e^- \rightarrow H_2+20H^-$	$2H^+ + 2e^- \to H_2$	$H_2 0 + 2 e^- \to H_2 + 40^{2-}$
Overall:	$2H_2 0 \rightarrow 2H_2 + O_2$	$2H_20\to 2H_2+O_2$	$H_2 O \rightarrow H_2 + \frac{1}{2} O_2$	$H_2 O \rightarrow H_2 + \frac{1}{2} O_2$

According to Table 3, PEMWEL has the fastest response time among the other types. Hence, PEMWELs find extensive use in power system applications, especially where a swift response is essential for perceiving system stability. This motivates the authors to focus on modelling this specific

type, thereby ensuring a more concise article. In this context, the reader is invited to check the following subsection for the PEMWEL mathematical model.

4. Electrical modelling of WELs

As mentioned earlier, the primary barrier to widespread commercialization of WELs in power systems is their capital cost. Thus, a precise and reliable assessment of their behaviour shall be implemented to identify any system defect before installing such expensive apparatuses [78]. Consequently, numerous researchers are seeking for properly deriving comprehensive mathematical models that can accurately simulate WELs' performance during various operating conditions [79, 80]. In fact, a robust and efficient model cannot only evaluate the WEL's response but also highlight the operational parameters that optimize and predict WELs' performance in a wide range of operating scenarios [81].

Given that WELs are multiphysics devices in which various chemical and thermal processes occur simultaneously during operation, deriving a single model to comprehensively describe all these phenomena is challenging. So, various electrical models have been introduced each varies depending on the deriving approach, the studied behaviour, and the mathematical formulation, as depicted in Fig. 5 [82, 83].



Fig. 5. Categorization of WELs' modelling

Among these approaches, the electrochemical model plays a significant role in accurately replicating the polarization characteristics (V-I curve) of PEMWELs [84]. This is achieved through the integration of chemical and physical parameters, including water content, hydrogen production rate, membrane material, and cell area [85]. As illustrated in the model's equivalent circuit shown in Fig. 6, the PEMWEL's input voltage is a resultant of a voltage source, representing the minimum required

potential to start the chemical reactions, and a series of operational losses. More precisely, these losses encompass activation and concentration overpotentials, each depicted by a capacitor-resistor branch, alongside ohmic voltage drop originating from membrane and connection resistances. The model's results are illustrated through a V-I curve, as depicted in Fig. 7 [78, 86].



Fig. 6. Equivalent circuit of the electrochemical model



Fig. 7. Typical polarization curve of PEMWEL

As a single WEL cell is inadequate for meeting the demanded hydrogen production, multiple cells are linked together serially forming a stack, ensuring the required hydrogen output. Thus, input voltage of the WELs' stack $V_{i/st}$ in (V) can be mathematically described by (1).

$$V_{i/st} = N_{elz} \times V_{i/elz} \tag{1}$$

where, N_{elz} is the number of serially connected cells. $V_{i/elz}$ is the cell input voltage in (V) which is formulated by (2) [87].

$$V_{i/elz} = E_{oc} + V_{ac} + V_{\Omega} + V_{cn} \tag{2}$$

where, V_{ac} , V_{Ω} , and V_{cn} symbolize the activation, ohmic, and concentration voltage drops in (V), respectively. E_{oc} represents the minimum required voltage to initiate the cell in (V) and is given by Nernst equation, as in (3) [86].

$$E_{oc} = 5.1822 \times 10^{-3} \left[\Delta G + 0.0821 \cdot T_{elz} \cdot ln \left(\frac{P_H \sqrt{P_o}}{P_W} \right) \right]$$
(3)

where, the change in Gibbs-free-energy is represented by ΔG in (kJ/mol). T_{elz} refers to the WEL's operating temperature in (k). P_H , P_O , and P_W point out the pressure of hydrogen, oxygen, and water in (bar), respectively.

To describe the initial sluggish of the chemical reaction, the activation overpotential V_{act} is computed by (4).

$$V_{act} = 4.2545 \times 10^{-4} \cdot T_{elz} \left[\frac{1}{\delta_n} \sinh^{-1} \frac{J_{elz}}{2J_n^{\circ}} + \frac{1}{\delta_{th}} \sinh^{-1} \frac{J_{elz}}{2J_{th}^{\circ}} \right]$$
(4)

where, δ_n and δ_{th} symbolizes the charge transfer coefficients of anode and cathode, respectively. The WEL's current density is denoted by J_{elz} in (A/cm^2) . The anode and cathode exchange current densities are indicated by J_n° and J_{th}° in (A/cm^2) , respectively.

Furthermore, the linear region in the polarization curve, caused by the ohmic losses, can be expressed by (5) [88].

$$V_{\Omega} = J_{elz} \cdot (R_m + R_{ex}) \tag{5}$$

where, R_{ex} refers to the resistance due to the external leads in $(\Omega. cm^2)$. The membrane resistance is represented by R_m in $(\Omega. cm^2)$ and is computed by (6) [86].

$$R_m = \frac{l_m}{\sigma_m} \tag{6}$$

where, l_m is the thickness of the WEL's membrane in (*cm*). The membrane's resistivity, donated by σ_m in Ω . cm^{-2} , is function of the WEL's temperature and the membrane's water content λ , as given by (7).

$$\sigma_m = (0.005139\lambda - 0.00326) \cdot \exp\left[4.1848\left(\frac{T_{elz} - 303}{T_{elz}}\right)\right]$$
(7)

At heavy current density, the chemical reaction encounters an impediment due to the high oxygen concentration at the membrane area. This is represented by the concentration voltage drop and expressed by (8).

$$V_{cn} = 4.2545 \times 10^{-4} T_{elz} \cdot ln \left(\frac{J_{max}}{J_{max} - J_{elz}}\right)$$

$$\tag{8}$$

where, J_{max} represents the maximum current density in (A/cm^2) . Finally, the produced hydrogen flowrate Q_H in (m^3/sec) is expressed by (9) [87].

$$Q_H = 4.2545 \times 10^{-4} \frac{J_{elz} \cdot T_{elz}}{P_H}$$
(9)

While the electrochemical model discussed thus far primarily addresses PEMWELs, it is also instructive to examine alternative electrolyzer technologies, such as ALWELs, AEMWELs and SOWELs, and how their modelling approaches differ in response to unique operational conditions. An ALWEL is typically described by a base thermodynamic voltage plus activation, ohmic, and concentration overpotentials. The ohmic loss is higher than in PEM systems due to the alkaline solution's lower conductivity and the diaphragm thickness. Activation overpotentials, stemming from electrode kinetics, can be captured via resistor–capacitor (RC) or resistor–inductor–capacitor (RLC) branches, while concentration overpotentials account for gas bubble formation and mass transport limitations [89].

AEMWELs resemble PEMWELs in having a reference voltage source and similar loss terms, but the higher ionic resistance of AEM membranes amplifies ohmic losses. Activation overpotentials are modelled by RC branches, with differences in catalyst composition. Concentration overpotentials can become significant if water supply or gas evacuation is restricted. Evolving models also consider membrane carbonation and degradation to predict dynamic performance [90].

SOWEL modelling must integrate electrochemical and thermal dynamics due to elevated temperatures that alter reaction kinetics and material properties. Multiple overpotential terms, including activation, ohmic, and concentration, are typically included, with separate RC or RLC branches for different electrode interfaces. Temperature-sensitive ohmic resistance arises from the electrolyte and interconnects, while mass transport effects may require diffusion-limited current expressions. Long-term factors like electrode delamination or electrolyte cracking are also included in advanced models [91]. Table 4 compares the primary parameters, challenges, and representative equivalent circuit elements across different electrolyzer types.

Item	m ALWEL PEMWEL		AEM	SOWEL	
Overpotentials	• Activation	 Activation 	 Activation 	 Activation 	
	• Ohmic	• Ohmic	• Ohmic	• Ohmic	
	Concentration	 Concentration 	• Concentration	• Electrode polarization	
Ohmic loss	• Diaphragm/electrolyte	• Membrane	• Higher	• High-temperature	
drivers	resistance.	conductivity	membrane	electrolyte/interconnect	
		(hydration- dependent).resistance resistancethan		resistance.	
Equivalent	• Voltage source	• Similar multi-	• Like PEM, but	• More complex.	
circuit	(thermodynamic).	branch RC	with higher	• Often featuring multiple	
structure	• Multi-branch RC (or	structure, with	ohmic resistance	RC (or RLC) branches to	
	RLC) networks for	emphasis on		capture temperature-	

Table 4. Comparative overview of modelling approaches across electrolyzer types

	activation and	membrane	due to OH-	dependent polarization
	concentration	resistance and	conduction.	effects.
	overpotentials.	water		
		management.		
Dynamic	• Moderate: gas bubble	• High: rapid load-	• Moderate to	• High: must integrate
modelling	formation and mass	following	high: evolving	thermal gradients,
complexity	transport influence	capability	designs require	potential material
	transient response.	requires time-	tracking	degradation, and transient
		varying	membrane	electrochemical
		membrane	carbonation and	responses.
		hydration	electrode	
		modelling.	kinetics.	
Modelling	• Ensuring accurate	• Capturing water	• Limited data on	• Elevated temperatures
challenges	representation of	management	membrane	accelerate
	diaphragm thickness, ion	(membrane	durability.	electrode/electrolyte
	transport, and bubble	hydration) and	• Higher ohmic	degradation
	formation.	load transients.	losses.	• Requires coupled thermal-
		• Catalyst	• Evolving	electrochemical
		degradation.	catalyst designs	modelling.
Representative	[89, 92]	[93, 94]	[90, 95]	[91, 96]
references				

5. Power electronics for electrolyzers and fuel cells applications

Modern power systems incorporate large amounts of RESs along with conventional power plants, with distinct volage profiles. Thus, the integration of WELs and FCs in such systems necessitates an interface to link the system voltage to their operating and generated voltages, respectively [20]. Specifically, the DC converter serves as such an interface since most RESs generate high DC voltage, whereas WELs need a low DC voltage input [21]. Conversely, FCs are employed as a standby power source that can rapidly inject electrical power to the system when the power supplied by RESs is insufficient. However, the injected power is delivered at a lower voltage than the DC bus. Hence, a DC converter is also obligatory to step up the FC's output voltage. Another crucial feature of WELs and FCs is their inherently non-linear electrical response. Hence, a DC converter is essential to regulate the voltage regardless of the loading conditions [14]. Fig. 8 offers a schematic explanation for the configuration of WELs and FCs with RESs in a practical power network [32]. Indeed, a proper DC converter for WELs and FCs applications shall meet certain features, like low expenses, high conversion ratio and efficiency, substantial power density, minimized output current ripples and electromagnetic interference, and capability to operate reliably during switching failure [20, 97].



Fig. 8. Schematic configuration of WEL/FC-based power network

Particularly, the non-isolated and isolated DC converters dominate the WEL and FC applications throughout the literature [20, 21, 98-100]. Thus, Table 5 presents a summarized comparison of the key features of each category. In addition, the following subsection introduces an overview of their topologies.

Converter's type	Non-i	solated DC converters	Isolated D	C converters
Index	Boost converter	Buck-boost converter	Flyback converter	Full-bridge converter
Electrical isolation	×	×	Via transformer	Via transformer
Conversion type	Step up	Step up/down	Step up/down based on the transformer turns ratio	Step up/down based on the transformer turns ratio
Power range	Low-medium	Low-medium	Low-medium	Medium-high
Complexity	Simple	Simple	Moderate	Complex
Efficiency	High	Moderate	Low	High
Cost	Low	Low-moderate	High	High
Electromagnetic compatibility	Low	Low	Moderate	High
Switching frequency	High	High	Moderate-high	High
Reliability	Vulnerable to single point of failure	Vulnerable to single point of failure	Moderate	High with reduced capacity

Table 5. Basic comparison of DC converters

5.1. Non-isolated DC converters

Principally, non-isolated converters are extensively utilized in numerous applications involving low and medium DC voltages due to their affordability, simplicity, compact design, and ease of control [101, 102]. Typical examples of such converters are the conventional buck-boost converter (BBC) [103], interleaved buck-boost converter (IBBC) [104], bidirectional buck-boost converter [105], and floating-interleaved buck-boost converter (FIBBC) [106].

Conventional BBCs are distinguished by their low cost and uncomplicated circuit connection. This augments their wide implementation in WEL-based power systems. However, such converters experience considerable ripples in output current, which can be mitigated by raising the inductor size or magnifying the switching frequency [102]. Nevertheless, increasing either of these parameters will inversely impact energy efficiency, as it will lead to higher core and switching losses. Another concern is that to attain a high step-up or step-down ratio, BBCs must operate at a significantly low duty cycle, which can potentially result in interrupted conduction mode. This directly affect the converter's output voltage connected to the WEL. Again, this issue is addressed by elevating the inductor rating leading to a higher loss, price, and size [103]. Moreover, conventional converters suffer also from reverse recovery power loss due to the diode's reverse recovery potential. Such loss is function of the initial current, operating temperature, and the change rate of the switching current [101]. Thus, a synchronous buck converter, an upgraded version of the conventional buck converter, is proposed in [107] to minimize the aforementioned loss. In this design, a power transistor replaces the freewheeling diode, as illustrated in Fig. 9.



Fig. 9. Circuit diagram of synchronous buck converter

For the sake of obtaining a high step-down ratio, a quadratic buck converter is proposed in the literature [108]. In this configuration, two buck converters are serially connected through a single power transistor, as shown in Fig. 10. Since this type runs only with a single power switch, the converter lacks fault tolerance. Also, the power transistor encounters a high potential stress throughout the operation. Thus, the authors in [109] have introduced multiquadric buck converters to minimize the potential stress across the switches. Another technique to increase the step-down ratio is to use the tapped-inductor buck converter [110]. In this design, the conventional inductor is replaced by a tapped inductor with a single primary and single secondary winding, as depicted in Fig. 11. By this architecture, the step-down voltage ratio is controlled by changing the turns ratio of the tapped winding n_p/n_s along with the duty cycle D, as described in (10). Nonetheless, the winding leakage flux between both windings causes spikes across the switch.

$$\frac{V_{out}}{V_{in}} = \frac{D}{D + (1 + n_p/n_s)(1 - D)}$$
(10)

For the same reason, a switched inductor-switched capacitor buck converter is developed in [111] to magnify the voltage stepping ratio, as shown in Fig. 12. However, this design cannot guarantee continuous operation in faulty conditions as it only includes one power switch.



Fig. 10. Circuit diagram of quadratic buck converter



Fig. 11. Circuit diagram of tapped-inductor buck converter



Fig. 12. Circuit diagram of switched inductor-capacitor buck converter

When aiming to amplify the delivered power, the IBBC and FIBBC are the best alternatives to the traditional BBC because the current is divided through separate legs. Basically, both configurations are formed by connecting multiple BBCs via a common DC bus. For instance, the 3-legs IBBC, whose circuit is shown in Fig. 13, offers optimal solution for minimizing output current distortion and magnifying energy efficiency [112]. Another merit is that the power delivered from/to a WEL/FC through IBBCs remains unaffected by the failure of a single power switch. Nevertheless, in the case of 3-legs IBBC, such a failure causes a 50% increase in current stresses on the unfaulty legs compared to the normal operation. This overstress, accompanied by excessive temperature, can severely affect the switches' reliability. Multi-leg IBBCs have also been studied in the literature for WEL/FC applications. For example, the authors in [113] designed a 1200W silicon-based Multi-leg IBBC for

FC operation. The results demonstrate a substantial reduction in current ripples by nearly 100%, along with an energy efficiency of approximately 90%, while operating at a switching frequency of 25 kHz.



Fig. 13. Circuit diagram of 3-legs IBBC

5.2. Isolated DC converters

Isolated DC converters are the optimal choice for high-voltage WELs/FCs, as they incorporate an intermediate AC domain utilizing a high-frequency transformer, as illustrated in Fig. 14 [100]. Particularly, the transformer plays a vital role in ensuring a low voltage input and a low current to the output side converter [114]. The AC phase is considered as an over current protection stage for the FCs [115]. Practically, it can be composed of various configurations including half-bridges [116], full-bridges [117], or multi-port interleaved bridges [118]. Conversely, a diode rectifier along with a multi-stage voltage source inverter represent the DC phase of such converters. Generally, flyback [119], half-bridge [120], full-bridge [121], forward [122], and push-pull [120] are among the commercial versions of isolated converters. Specifically, the literature claims that the best appropriate types for FCs are the half and full-bridge converters due to their galvanic isolation, the high frequency of the transformer minimize its size, their high voltage stepping ratio, and their smooth switching cycle that improves their efficiency [122].



Fig. 14. Schematic configuration of isolated DC converter

For example, an isolated half-bridge DC converter engaged with a smooth switching technique is introduced in [123] for an WEL application to diminish the switching losses, as captured in Fig. 15(a). In fact, this configuration is characterized by regulation simplicity, and applicability to achieve significant voltage ratio which is a must for WEL/FC systems, as described in (11).

$$\frac{V_{out}}{V_{in}} = \frac{D}{2n} \tag{11}$$

where, n is the transformer turns ratio.

On the other hand, it experiences high power loss due to the large turns ratio of the transformer which causes high magnetic flux resulting in excessive leakage reactance. Besides, it lacks the ability to operate in the event of a switch failure. Furthermore, the reader can browse [124] for investigating the full-bridge version utilized in WEL systems whose circuit diagram is depicted in Fig. 15(b). Primarily, its main advantage over the half-bridge isolated converter is its superior conversion efficiency. Moreover, a filter can also be integrated with it to eliminate current spikes on the power electronics.



Fig. 15. Circuit diagram of isolated DC converters

6. Control methodologies of WELs/FCs DC converters

An essential factor in optimizing WEL/FC-based systems is the converter control methodology, which ensures precise and efficient DC converter operation. Controllers regulate output voltage and current, mitigate power oscillations, and prevent adverse effects on WEL/FC performance [21]. Moreover, optimal control design enhances energy efficiency, minimizes converter losses, and extends component lifespan. To reiterate, buck converters step down DC bus voltage to power WELs, while boost converters raise FC output voltage to the required DC level [84]. For clarification, Fig. 16 illustrates the integration of controllers with DC converters. Control methods vary by technique, including analytical (e.g., PID, root locus, lead-lag compensators), model-based (e.g., sliding mode,

fractional order), and data-trained approaches (e.g., neural networks, fuzzy logic, model predictive control) [32]. However, only specific types have been recognized in the literature as effective tuners for DC converters in WELs/FCs-based power systems. Thus, the following subsections offer discussive illustrations on such types.



Fig. 16. Generic configuration of DC converter's controllers in WEL/FC-based networks

6.1. Proportional-integral controller

The proportional-integral (PI) controller is widely used due to its effectiveness and simple mathematical formulation. It operates by minimizing the deviation between the actual and desired values, combining two key components: the proportional term (k_p) , which adjusts the output based on the current error, and the integral term (k_i) , which accumulates the error over time to eliminate steady-state deviations [21, 84]. Accordingly, several studies have demonstrated the effectiveness of PI controllers in regulating the output voltage of DC converters, ensuring that WELs receive a consistent and appropriate power supply, even when the grid voltage fluctuates [32, 125].

In this regard, the authors in [126] proposed a novel PI tuner for an ALWEL-based network, as shown in Fig. 17(a). It's worth indicating that V_{el} , I_{el} , P_{rf} , and I_{rf} represent the WEL's actual voltage and current, as well as the desired preset power and current, respectively. On the other hand, R_{cn} and L_{cn} denote the buck converter's resistance and inductance, respectively. Additionally, they developed a tuning technique for the PI controller's parameters of the ALWEL, eliminating V_{el} by assuming it remains constant throughout the entire operation. They disregarded the internal dynamic response of the WEL by treating it as a linear resistor. Furthermore, an average switching approach is introduced for the WEL current control loop, as illustrated in Fig. 17(b). Finally, the authors assert that their design enhances the controllability of the DC link voltage by implementing the WEL current control as an inner control loop within the outer DC voltage control loop.



(b) Small signal current control loop of WELFig. 17. PI controller for WEL-based systems

Another contribution, as presented by the authors in [127], involved the development of two cascaded PI controllers to refine the current control signal more smoothly. Essentially, the inner loop regulates the WEL's current based on the reference signal generated by the outer loop, as revealed in Fig. 18. They employed the state-space representation to derive the transfer function of the WEL.



Fig. 18. Series PI controller for WEL-based systems

In contrast to previous efforts, the authors in [128] have taken into account the actual dynamic performance of the ALWEL. They applied control theory to model the WEL by using a step input and subsequently derived the corresponding second-order transfer function. Their proposed model is considered as an electrical circuit where it includes inductance L_{el} and capacitance C_{el} . These parameters represent the time delay due to a step input applied to ALWEL. However, L_{el} may be assumed zero for PEMWEL since it has a swift response compared to ALWEL. Physically, C_{el} defines the double-layer capacitance, R_m symbolizes the mass transport losses, and R_i refers to the nonlinear losses due to the material and external leads. The aforementioned state-space modelling, the proposed electrical model incorporates two state variables that characterize the controller's response, as described in (12) and (13) by neglecting R_{cn} .

$$(L_{el} - L_{cn})\frac{\mathrm{d}I_{el}}{\mathrm{dt}} = dV_{in} - V_{el} \tag{12}$$

$$C_{el}\frac{\mathrm{d}V_{dl}}{\mathrm{d}t} = I_{el} - \frac{V_{dl}}{R_m} \tag{13}$$

where, V_{in} and V_{dl} are the input voltage of the converter and the EL's electrical double layer voltage in (V), respectively.

As a result of (12) and (13), two PI controllers are necessary for both the outer voltage control and the inner current control in ALWEL. Nevertheless, only a single PI controller is employed in PEMWEL, as the electrical double layer is nearly neglected, thereby eliminating the need for the outer voltage control loop, as shown in Fig. 19.



Fig. 19. PI controller for ALWEL based on control theory

6.2. Sliding mode controller

Sliding mode control (SMC) is a nonlinear control technique that operates by forcing the system to follow a desired trajectory, known as a sliding surface or sliding manifold, in the state-space. The system is driven toward this surface using a high-gain control law, where it slides along the surface toward the desired equilibrium point, as revealed in Fig. 20(a). The main principle behind SMC is to switch control actions based on the system's state in relation to the sliding surface, allowing the system to converge to its target behaviour despite uncertainties or disturbances [129].

One of SMC's most significant advantages is its ability to handle systems with nonlinearities, unknown dynamics, and external disturbances, as concluded from its conventional structure seen in Fig. 20(b). However, SMC typically experiences chattering caused by the high-frequency switching in sliding mode, which can result in mechanical vibrations or excessive switching in power electronics [43,58]. Thus, recent SMC variants have been developed, like higher-order sliding mode control (HOSMC), terminal sliding mode control (TSMC), and super-twisting sliding mode control (STSMC) [130, 131]. In WEL-based systems, SMC provides precise control of DC converters by optimally regulating input voltage and current, maintaining efficient hydrogen generation. Thanks to its swift dynamic response,

SMC can ensure system stability despite load variations and disturbances. Moreover, its capability to deal with heavy nonlinear systems, like WELs, magnifies its robustness in sustaining the desired operating scenarios. In this context, the reader is invited to visit [132] for investigating practical applications of SMC in WEL/FC-based systems.

6.3. Backstepping controller

Principally, backstepping control offers a systematic design approach that utilizes a series of virtual controllers to stabilize complex systems. Specifically, the backstepping controller is constructed step by step using a recursive procedure, which ensures stability at each stage. Hence, it outperforms conventional techniques in managing nonlinear behaviours and unwanted disturbances [133]. Accordingly, backstepping control has several applications in DC converters, where it accurately controls output voltage and current to minimize energy loss and enhance stability over a wide range of loading conditions [134]. Furthermore, due to its capability to handle abrupt system changes and parameter uncertainties, backstepping control can address additional integrated control targets, such as power management and current limitation. Thus, the backstepping tuner has been employed to address various control challenges in FC-based power systems. For example, a backstepping controller is proposed in [135] to ensure smooth tracking of the maximum allowable power generated by the PEMFC, thereby extending its operational lifetime.



(a) Typical trajectory of a standard SMC



(b) Practical steps for designing an SMC

Fig. 20. Basic structure and concept of SMC

6.4. Model predictive controller

The popularity of Model Predictive Control (MPC) in WEL/FC-based power systems arises from its effectiveness in managing multivariable control systems with constraints and optimizing performance in real-time applications [136]. Particularly, MPC can predict the future response of WEL/FC operating variables, such as voltage, current, and hydrogen production/consumption, by utilizing their dynamic model. It updates the control inputs based on a continuously optimized objective function, subject to practical constraints, such as power and hydrogen production/consumption boundaries. This promotes MPC in applications involving WELs/FCs linked to RESs, where unpredictable variations in power supply may occur. Its robustness and precision in predicting system responses within preset boundaries contribute to maximizing the efficiency of WELs/FCs, thereby extending their operational lifetime. Moreover, MPC can adapt to additional control objectives, including hydrogen storage limitations, downstream applications, and power network stability [137, 138].

Practically, an MPC comprises four principal phases: prediction technique, optimization methodology, receding horizon, and feedback. The prediction technique is employed to foresee the system's upcoming states, while the control task is handled by the optimizer to diminish the fitness function without violating the system's constraints. The receding horizon approach implements only the first control action from the optimized sequence, advancing the horizon with each time step. Finally, the model is updated according to the latest measured data using the feedback strategy. Fig. 21 illustrates the basic block diagram of a practical MPC, showing how these phases work together to achieve efficient control [139].



Fig. 21. Schematic block diagram of MPC

6.5. Fuzzy logic controller

Owing to its ability to address the vagueness and inaccuracy inherent in sophisticated dynamic systems, fuzzy logic controller (FLC) has recently recognized as a potent control method in WEL/FC-based systems. In contrast to conventional controllers, FLC doesn't necessitate a precise mathematical model of the system, rather, it utilizes human-like reasoning to derive decisions predicated on imprecise information [140]. Fig. 22 showcases the three primary phases of implementing an FLC: fuzzification, rule evaluation, and defuzzification. Initially, it transforms distinct inputs, such as error signals, into fuzzy sets that denote varying degrees of membership within linguistic categories. Subsequently, it analyses these inputs through the application of specific rules to ascertain the suitable control action. Ultimately, the fuzzy output is reverted back to a distinct control signal. This methodology is particularly efficacious for the management of intricate, nonlinear systems without the necessity for exact mathematical models [141].

In WEL-based systems where variables such as input power, voltage and current can fluctuate due to RESs, FLC is of great advantage. It allows smooth control of the operation of the WEL by adjusting parameters to maintain optimal hydrogen production under changing conditions. FLC is perfect for online applications because of its adaptability in handling nonlinearities and disturbances, which guarantees the WEL runs effectively and within safe bounds. To further improve system performance and reliability, FLC can be easily integrated with other control strategies, such as PI or MPC. Because of its versatility and durability, FLC is a preferred option for handling the complexity of WEL-based power systems, especially when integrating RESs [142].



Fig. 22. Schematic block diagram of FLC

6.6. Neural network controller

Without any loops, a feedforward neural network (FNN) processes data in a single direction, from input to output. It is made up of an input layer that takes in data, one or more hidden layers that use activation functions and weighted transformations, and an output layer that generates the final prediction. In order to reduce errors during training and enable the network to learn intricate patterns and produce precise predictions, the network uses backpropagation to modify its weights. Over time, the network's performance can be enhanced and generalized through this process [143]. FNNs are employed to model and regulate the challenging behaviour of WELs/FCs because they resemble the human brain's capacity for learning and generalization from data. In these systems, particularly when connected RESs, FNN learn the best control strategies to can for managing variables like voltage, current, and hydrogen mass flowrate under variable load and power conditions. For example, a novel FNN-based maximum power point tracking controller is proposed in [144] to optimally tune the duty cycle of the DC converter connected to a PEMFC stack. The FC's voltage V_{fc} and current I_{fc} serve as the input data, while the converter's duty cycle represents the output. Fig. 23 illustrates the proposed FNN structure. The integration of the FNN with the DC converter ensures zero current ripple, optimizing the performance of the system.



Fig. 23. Schematic architecture of FNN

6.7. Adaptive neuro-fuzzy inference controller

Fuzzy logic and neural networks are combined in an adaptive neuro-fuzzy inference system (ANFIS), which allows it to learn and adjust to changing operating conditions. It functions by transforming input data into fuzzy sets, assessing relationships, and optimizing the membership functions and rules using neural network learning. ANFIS modifies these settings during training in order to reduce errors. Lastly, it transforms fuzzily output values into sharp ones [145]. ANFIS is an optimal solution for dynamic and nonlinear systems because of its integration of neural learning and fuzzy logic, which enables it to adapt over time. Because of its flexibility, ANFIS can continuously modify control parameters to achieve the highest efficiency levels, thereby optimizing the performance of WEL/FC. ANFIS can be used in FC systems to control output power and increase cell lifetime, and in WEL systems, it aids in maintaining ideal hydrogen production rates under a range of load scenarios [146].

Finally, Table 6 offers a comprehensive comparison of the previously discussed control techniques in terms of operation theory, stability, learning and adaptation capabilities, ability to handle nonlinear systems, tuning effort, computational burden, and suitability for real-time applications. Furthermore, an extensive summary of the recent state-of-the-art literature on the practical application of these techniques in WEL/FC applications is thoroughly covered in Table 7.

Controller's type	PI	SMC	Backstepping	MPC	FNN	ANFIS
Index						
Control concept	Linear, proportional- integral feedback control	Discontinuous, nonlinear, sliding surface control	Recursive, nonlinear control	Optimization-based, predictive control	Neural network-based learning and prediction	Combination of fuzzy logic and neural networks
Nonlinear systems	Limited capability	Excellent for highly nonlinear systems	Excellent	Can manage nonlinearities via model-based prediction	Can approximate nonlinearities with sufficient training	Effectively deal with nonlinearities via hybrid annroach
Tuning burden	Relatively easy, requires tuning of P and I gains	Requires designing the sliding surface and control law	Moderate, involves recursive design for each subsystem	Difficult, requires dynamic model and constraints tuning	High, involves designing and training the neural network	High, involves tuning both fuzzy rules and neural network
Stability	Moderate robustness, affected by parameter variations	Very robust to system uncertainties	High robustness due to recursive stability checks	Robust when models and constraints are well-defined	Limited robustness, dependent on training data	Good robustness, adapts to uncertainties
Computational complexity	Low, easy to implement	Moderate to high, especially with chattering reduction	Moderate, step-by-step design increases complexity	High, due to solving optimization problems in real-time	High, requires training and backpropagation	High, requires both fuzzy system design and neural network training
Chattering	No	Prone to chattering but can be mitigated with modifications	No	No	No	No
Real-time application	Suitable	Challenging due to chattering and high-frequency switching	Suitable with proper design	Requires powerful processors for real-time implementation	Using fast processors and good training	Possible but computationally challenging
Learning ability	No	No	No	No learning, but adapts by predicting future response	Learns through data training	Adaptive via learning from data and fuzzy rules
Performance due to disturbances	Moderate, may require additional tuning	Excellent, due to high robustness	High, effectively compensates disturbances	Good, with proper model and constraint definition	Depends on training data; may require retraining	Good, adapts to disturbances through learning

Table 7. Summary of recent literature in control techniques for practical WEL/FC applications

	Remarks	The proposed integration of the buck and full-bridge converters effectively achieves a high conversion ratio, making it suitable for medium and high-power ELRs. Additionally, the PI controller successfully eliminates current ripples and maintains stability during studen load variations.	The authors propose a novel MPPT for a standalone PV system using a PI-improved FLC. The suggested controller outperforms conventional incremental conductance- based MPPT by extracting the maximum PV power. Additionally, it ensures a stable DC bus voltaee. recardless of variations in solar irradiance or load conditions.	A two-level control methodology is introduced for optimal energy management and MPPT of the RESs, while also stabilizing the DC bus voltage by regulating the power flow of the battery, FC, and WEL. The proposed HOSMC achieves a minimal steady- state error of 0.005% and an overshoot of 0.07%.	A modified STSMC is designed for optimal wireless power transfer by supressing windup impacts due to integral deviation accumulation and smoothing the transition	
	Control method	Id	PI-enhanced FLC	HOSMC	STSMC	
erter	DC bus voltage (V)	150- 220	37	600	400	20
DC conv	Type	Buck and full bridge	Buck-boost	Traditional buck	Traditional buck	
FC	Capacity (kW)	×	×	40	13	
	Type	×	×	PEM	MN	
VEL	Capacity (kW)	0.4	MN	50	1.5	
1	Type	PEM	MN	MN	MN	
	EST	×	BESS	BESS	BESS SESS	
	RES's type	WT	ΡV	WT Tidal	×	
	Grid mode	Off	Off	Off	Off	
	Ref.	[147]	[148]	[149]	[150]	

Table 6. Detailed comparison of recent controllers in WEL/FC systems

78

to the desired current values. The grey wolf optimizer is utilized for optimal extraction of the controller's parameters. The proposed control method has the minimum settling and rise times of 0.4356 s and 0.0364, respectively. An upgraded version of global TSMC-based energy management system is developed based on the battery's state-of-charge and hydrogen level in the tank to enhance the battery and FC lifetime over diverse operating conditions. Moreover, the controller affirms its capability to regulate the DC bus due to sudden load or weather chances.	The authors address the indirect coupling of PV and WEL through a DC converter using SMC to optimally tune its duty cycle. The findings reveal that energy utilization ramps up to 77.6% when employing the SMC for tuning the converter's duty events.	unity of the second second second for optimal managing the power flow of grid- tin this attempt, an SMC is employed for optimal managing the power flow of grid- connected WT along with WEL, FC, BESS, and SESS seeking a stable power supply system for electric fraction amplications.	A novel application of adaptive between the DC for voltage regulation of the DC converter is addressed. Numerically, at unexpected variation of operational scenarios, the controller attains the lowest settling time and percentage overshoot of 0.05 and 17 5%.	The authors elaborate on designing an integral backstepping controller for extracting largest allowable PV power, stabilizing the common bus voltage, and controlling the battery's state-of-charge. For further validation, the stability of the proposed controller is evaluated using Lyanunov method.	This work proposed a subscription manual of a standalone system comprising PEMFC stack feeding adjustable resistive load through buck/boost DC converter. The suggested controller significantly minimizes chattering of the traditional one, ensuring a stable moregion researdless the inierced distributions.	This article introduces an innovative distributed mathematical modelling of the DC converters controlled by three separated controllers for optimal techno-economic operation of PV, FC, and WEL. Furthermore, the authors proposed a communication strategy for the three controllers to achieve supply-demand power balance, regulated DC bus voltage.	A two-layer control approach is addressed in this work for a sustained power flow to the loads and an ensured high-quality for the microgrid's variables. Specifically, the MPC represents the supervisory control layer that determine the desired values for the primary control layer leading to zero error in the reference values. Besides, it ensures that FC, BESS and WEL operate safely within their practical constraints for evending their lifetime.	The proposed controller of the stack interleaved buck DC converter precisely allocate the optimal power delivered to the PEMWEL along with stabilizing the converter output voltage and minimizing its output current ripples. Moreover, the authors have addressed the state space modelling of the converter, along with the dynamic	movening of the convenent. An isolated microgrid comprising PV, along with ESTs, supplying a DC load and hydrogen production system. A PI controller is utilized to regulate the inner current control loop.	This attempt reveals optimal design of a PI controller relying on the small signal analysis of the three-level interleaved buck converter, enhanced by experimental setup. The introduced work ensures an accurate control of the produced hydrogen quantity for optimal DC bus voltage balancing in the presence of RESs.
TSMC	SMC	SMC	Backstepping TSMC	Integral backstepping	Backstepping SMC	Distributed economic MPC	MPC	Double-loop PI	PI	Id
700	37	400	100- 200	30	12	300	230	100- 200	80	100- 200
Traditional buck	Quadratic buck	Traditional buck	Traditional buck	Dual input triple output buck	Buck/boost	Buck/boost	Bidirectional	Stacked interleaved buck	Traditional buck Interleaved	Three-level interleaved buck
13	×	MN	×	MN	100	10	-	×	1	×
PEM	×	MN	×	Virtual	PEM	PEM	PEM	×	PEM	×
10	MN	MN	0.4	×	×	10	-	0.4	1.26	0.4
MN	PEM	MN	PEM	×	×	AL	AL	PEM	PEM	PEM
BESS	×	BESS SESS	×	BESS	×	×	BESS	×	BESS SESS	×
WT	ΡV	WT	×	Ы	×	PV	PV WT	PV WT	ΡΛ	×
Off	Off	On	Off	ÛĤ	Off	Off	Off	Off	ĴĴO	ÛÎ
[151]	[152]	[153]	[154]	[155]	[156]	[157]	[158]	[159]	[160]	[161]

A novel control methodology is presented for tuning four-phase interleaved boost converter linked to PEMFC stack. Specifically, it involves backstepping controller for the inside loop, while STSMC is for the outside loop. The designed approach guarantees a stabilized DC bus voltage within diverse levels, robust capability of withstanding strong sudden disturbances, and effective suppression of the FC's current ripoles.	An innovative modelling of a grid-connected offshore hybrid RESs-based network is thoroughly discussed for optimal power dispatch between the grid and PEMWEL for cost-effective hydrogen generation. The MPC is applied to the WEL's buck converter, based on the allocated hydrogen production strategy, for optimal economical hydrogen production.	The proposed methodology engages the energy allocation system with the local feedback control for an optimal individual controller. The findings reveal that the centralized economic MPC outperforms the others in attaining the minimum rout mean square error between the desired and actual powers of 0.1267.	The authors introduce a novel sizing technique involved with optimal energy management system based on an economic MPC. The studied problem is considered as a dual-level optimization task comprising an inner and outer control loop. Particularly, the optimal sizing of the RESs is tackled by the outer loop, while the inner loon optimally manages the energy flow for each component.	In order to overcome the inaccurate prediction of loads and RES's share and the slowness of the programming techniques over large timescale, the authors propose an energy management approach based on MPC and dynamic programming concepts. The outcomes show that a balanced power flow is achieved, the FC and BESS operate within rated power scales, and the error signals are almost brought to zero.	An FNN is proposed to optimally control the wind output power and the battery's state-of-charge by tunning the FC and WEL outputs according to the battery's parameters, such as charging/discharging energy. The suggested FNN controller optimally manages the battery's state-of-charge leading to improving its lifetime and significantly diminishes the transient time.	The key contribution of this work is utilizing a three-level interleaved buck DC converter which has an electrical withstand capability against switches failure, high voltage step-down ratio, and insignificant output current ripple. Additionally, the proposed controller experiences a swifter response and higher stability in case of narameters' uncertainties with research othe conventional PI controller.	The authors proposed a control methodology base on ANFIS for the operation of a hybrid AC/DC microgrid. The DC bus is connected to PV, FC, and WEL, while the WT and BESS are linked to the AC bus. An interconnected converter is placed between the two buses to control the power exchange according to the DC bus voltage and AC bus frequency. The suggested controller succeeds in reducing the output power and voltage transients of the BESS and minimizing the voltage deviations at the AC bus at the nodes.	The authors introduce an ANFIS-based methodology for controlling bidirectional buckboost DC converters in an isolated DC microgrid. The findings signify the superiority of ANFIS compared to PI and FLC controllers in terms of achieving the minimum voltage overshoot of 10%.
Backstepping STSMC	MPC	Centralized economic MPC	Economic MPC	Dynamic programming MPC	FNN	Improved SMC	ANFIS	ANFIS
26	500	70	MN	WN	MN	100- 200	800	500
Four-phase interleaved boost	Traditional buck	Traditional buck	Traditional buck	Traditional buck	Traditional buck	Three-level interleaved buck	Traditional buck/boost	Bidirectional buck/boost
0.5	×	×	-	n	MN	×	10	9
PEM	×	×	PEM	PEM	MN	×	MM	PEM
×	10000	10	-	ς	MN	0.4	10	×
×	PEM	AL	PEM	PEM	MN	PEM	MN	×
×	×	×	BESS	BESS	BESS	×	BESS	BESS
×	PV WT	PV	PV WT	PV	W	ΤW	PV WT	PV
Off	On	Off	Off	O	Off	Off	Off	Off
[135]	[162]	[163]	[164]	[165]	[166]	[167]	[168]	[169]

Alongside modelling the dynamic behaviour of PEMFCs, the authors propose an	ANFIS-based MPP1 tuner for a DC microgrid. The addressed controller optimizes the converter's output voltone and current for enhancing the voltone stability.
ANFIS	
300	
Conventional	boost
1	
PEM	
×	
BESS ×	
ΡV	
Off	
[170]	

7. Power system integration of WEL-based systems

7.1. WEL's applications in modern power systems

The rapid expansion of RESs in modern power systems has raised stability concerns due to their intermittent nature, especially in PV and WT systems [3, 171]. Unlike conventional power plants, RES output fluctuates with weather conditions, causing frequency and voltage instability [172, 173]. On the other hand, WELs offer a dual benefit, producing green hydrogen while providing ancillary grid services. By adjusting their operation, they help balance supply and demand, regulate frequency and voltage, and enhance grid resilience [26, 29-32]. Thus, the following subsections explore these services, emphasizing their role in optimizing power system performance and supporting a sustainable energy transition.

7.1.1. Frequency regulation

Maintaining a stable frequency is crucial for reliable operation and protecting connected equipment. Frequency fluctuations occur when supply and demand are unbalanced, causing deviations from the nominal 50 Hz or 60 Hz [174]. In response to these alternations, WELs help stabilize frequency by adjusting their power consumption. During over-generation, they increase hydrogen production by consuming excess electricity, reducing frequency to normal levels. Conversely, during undergeneration, they decrease power consumption, freeing capacity to balance the grid, as shown in Fig. 24 [175, 176].

Moreover, WELs are scalable devices capable of adapting to several network configurations and sizes. For instance, in distribution systems where the integration of RESs is rapidly increasing, small-scale WELs can be deployed to provide secondary frequency control. Accordingly, an online simulation of engaging a generic front-end controlled 120 kW WEL for providing frequency compensation is reported in [29]. The findings highlight the capability of the PEMWEL to provide an immediate and appropriate response, thereby significantly reducing frequency errors. On the other hand, placing larger-scale WEL systems at key locations such as transmission hubs or substations can significantly enhance grid stability and frequency control over a larger area. By optimal situating these WELs at specific points in the transmission network, they can effectively stabilize grid frequency and address larger-scale disturbances. Their rapid response to varying grid conditions aids in mitigating fluctuations and reinforcing overall system reliability. In this context, the authors in [177] conducted an efficacy assessment of utilizing PEMWELs in conjunction with PEMFCs for primary frequency control. This evaluation was based on accurate forecast data derived from an actual grid scenario in

the Netherlands for the year 2030. The outcomes affirm that PEM devices can effectively compensate frequency deviations with respect to conventional generators under identical reserve capacity.

However, there are major obstacles that hinder the huge penetration of WELs in power systems. For instance, the swift dynamic response still only characterizes specific WEL's types, such as PEMWELs [30]. Furthermore, the WELs' capital cost of installation and maintenance are still high. Thus, numerous research attempts are trying to tackle these problems individually [22].



Fig. 24. WEL's contribution to frequency regulation

7.1.2. Voltage regulation

WELs also play a key role in voltage regulation, ensuring stable grid operation [178]. While traditionally considered DC loads consuming only active power, advancements in power electronics, such as grid-forming inverters, now enable WELs to control both active and reactive power [128, 179]. By adjusting inverter settings, WELs can absorb reactive power to lower voltage during overvoltage conditions or inject reactive power to raise voltage when needed. This regulation is achieved without disrupting hydrogen production, enhancing grid stability and flexibility [29, 32, 180].

Consequently, the authors in [181] propose a novel fleet control of grid-connected WELs to mitigate the penetration effect of PV into distribution systems. Their outcomes signify how the implemented fleet control can effectively reduce voltage oscillations, mitigate overvoltage issues, and stabilize the intermittent operation of control devices caused by the discontinuous nature of PV generation. This enhances overall grid stability and ensures smoother operation in systems.

Furthermore, the authors in [182] have fully defined the electrochemical model of a 0.5 MW AWEL involving its inherent dynamic response and nonlinearity. The studied WEL was connected to the grid through a phase-shifted full-bridge active front-end converter. The ultimate goal was to mitigate the voltage harmonic distortion and optimize the system behaviour under diverse operating conditions

including severe voltage deviations. According to the simulation outcomes, the proposed system successfully minimized the harmonic contents to match the Danich electricity grid code.

Another attempt is presented in [183] for analysing the gride code requirements for linking mega-scale WEL to the power network. Specifically, the authors have introduced a detailed model for a grid-integrated WEL via triple-phase h-bridge converter. Moreover, they developed a low voltage ride-through methodology. The integrated system has proved its efficacy in supplying sufficient reactive power to compensate voltage deviations.

7.1.3. Network balancing

Basically, network balancing ensures equilibrium between electricity supply and demand [29]. WELs contribute by acting as flexible loads, adjusting power consumption based on grid conditions. During excess generation from PV and WT, WELs increase hydrogen production, preventing renewable curtailment and reducing grid overload [180]. On the other hand, during high demand or low RES output, WELs reduce or pause consumption, freeing power for critical loads and easing grid strain [184]. This flexibility helps stabilize supply-demand balance while enabling hydrogen storage for later energy or industrial use, enhancing profitability, resilience, and renewable integration [185].

In this regard, the authors in [186] discussed how WEL plants can maximize the profit share of a WTdriven hydrogen production system interconnected to gride services. Particularly, they simulated practical grid services along with actual data of the energy price and wind energy profile from certain countries like Norway, France, Spain, and Italy. The findings highlight the capability of WELs to maximize economic profit through participation in applicable grid services by utilizing the unexploited capacity of generated hydrogen.

7.1.4. Grid congestion

Grid congestion occurs when transmission or distribution networks are overloaded, restricting efficient power flow, especially in high-RES areas where infrastructure struggles to handle peak generation. Traditional solutions focus on grid expansion and modernization, while smart grids with ESTs offer real-time monitoring, demand-side management, and enhanced RES integration, providing a flexible approach to congestion management [3, 16, 29, 173].

WELs help alleviate congestion by acting as controllable loads in congestion-prone areas. They absorb surplus electricity, converting it into hydrogen for storage, preventing bottlenecks and renewable curtailment. Their dynamic operation allows real-time energy flow control, reducing grid strain while maximizing RES utilization. Additionally, WELs lower the need for costly infrastructure upgrades, making green energy integration more efficient and economically viable [128, 187]. In this context, the authors in [188] conducted a techno-economic assessment of a GHP system connected to a medium-voltage grid, supplemented with RESs. The case study was implemented in the Netherlands, considering price-aware electricity customers who adjust their consumption based on fluctuating electricity prices. Besides, the proposed economic-incentive optimization methodology involves single leader, grid utility, and several followers, including RESs' owners and electricity consumers. Specifically, the utility allocates the time-variant congestion tariffs at each individual busbar, accordingly, the followers adjust their power generation or dissipation to minimize the fitness function, which is the operational cost. Herein, WELs demonstrate their ability not only to adapt to dynamic pricing strategies but also to contribute to the grid's profit share. By responding to fluctuating electricity prices, WELs can optimize their operation, consuming energy when prices are low and generating hydrogen efficiently.

For the same purpose, the authors in [189] discussed how WELs could effectively support the network management, configuration, and congestion in Germany. They proposed a generalized optimization approach comprising electricity price's variations and congestion management for diverse operating modes and strategies. Particularly, different WELs' operating scenarios, distinct WELs' installed capacities, and various industrial carbon-neutrality strategies are comprehensively evaluated. The outcomes underscore that WELs demonstrate superior performance in addressing power congestion issues, particularly when implemented within a distributed-industries framework.

7.1.5. Black start

Black start refers to restoring power to a grid after a shutdown without external energy sources. Conventional power plants require external power to restart, making grid re-energization challenging. Black start-capable units, typically small self-sufficient generators, play a key role in this process [190]. Although WELs are still underexplored for black start, they can serve this function when coupled with hydrogen storage and FCs. Stored hydrogen can be converted back into electricity, supplying the initial power needed to restart larger generators and gradually restore the grid. Their independent operation, quick response, and sustainability make WELs valuable for grid recovery. As RES penetration increases, reliance on fossil-fuel-based black start resources may become impractical, positioning WELs as a green and efficient solution for power system restoration [191].

A systematic summary of recent attempts to utilize WELs in various power system applications is presented in Table 8. Several assessment criteria have been used to evaluate the effectiveness of WEL control methodologies in achieving goals such as frequency and voltage regulation, power balancing, and grid congestion management. Commonly employed criteria include the integral absolute error (IAE), integral time absolute error (ITAE), integral square error (ISE), integral time square error

(ITSE), and mean square error (MSE), all of which measure the deviation between reference and actual values for frequency, voltage, or power.

7.2. Advanced grid strategies for hydrogen integration

As concluded from the previous subsection, GHP via WELs presents significant benefits for power systems in terms of long-term energy storage and ancillary services. However, fully realizing these advantages requires not just technological innovation in WELs but also advanced grid management. Consequently, this section explores how flexible network topology, dynamic thermal rating (DTR), and other distributed energy resources (DERs) can synergize with GHP to enhance overall system reliability and resilience.

Flexible network topology adjusts circuit breakers and tie lines in real time to redistribute power flows resulting in congestion reduction and greater renewable penetration [192, 193]. Furthermore, DTR raises transmission limits based on real-time ambient conditions, allowing operators to channel otherwise-curtailed renewable energy toward WELs. This reduces hydrogen costs by boosting WEL utilization and deferring major infrastructure upgrades. Accordingly, short-term measures like topology reconfiguration and DTR pair effectively with hydrogen's longer-term buffering capacity which create a robust approach to handling intermittent RESs.

Other DERs, such as batteries, offer high-power but short-duration buffering, while hydrogen excels at longer timescales [194]. Hence, coordinating these technologies helps manage both immediate fluctuations and prolonged renewable deficits. For instance, batteries stabilize frequency or voltage quickly, while hydrogen addresses seasonal imbalances. On the other hand, DTR optimizes power flow so that grid operators can dispatch smoothly excess wind or solar to WEL.

Case studies show that network reconfiguration and DTR can jointly reduce wind curtailment, bolster grid stability, and improve economic returns [192-194]. Achieving this synergy, however, demands investments in sensor technology, real-time monitoring, and advanced control algorithms. Thus, ongoing research targets multi-objective optimizations that integrate WEL dispatch with grid reconfiguration to ensure cost-effectiveness and decarbonization without compromising reliability. As technology matures and costs decline, coupling hydrogen with flexible grids is poised to become a key pillar of future resilient power systems.

7.3. Hierarchical and distributed energy management in hybrid grids

The integration of GHP into modern power systems requires advanced energy management strategies to optimize efficiency and ensure stability. Thus, hierarchical and distributed energy management frameworks can play a key role in coordinating WELs, energy storage, and DERs in hybrid AC/DC grids [195, 196]. Particularly, hierarchical energy management operates across multiple levels. At the

supervisory level, long-term hydrogen production is optimized based on market conditions and load forecasts. The network management layer balances AC/DC power flows, WELs, and storage assets to improve grid flexibility. Finally, device-level control autonomously adjusts power converters and WEL operation in real time [197]. This structured approach can optimize energy use, reduces congestion, and enhances grid resilience.

In contrast, distributed energy management shifts decision-making to interconnected control nodes, enhancing scalability and adaptability. Using cloud-edge-device cooperation, cloud intelligence performs system-wide optimization, edge computing manages local grid conditions, and device controllers can adjust hydrogen production in response to real-time demand [196]. This approach can improve direct DC coupling for WEL, reduces conversion losses, and ensures efficient RESs utilization.

A key advantage of hierarchical and distributed energy management frameworks is their ability to coordinate hydrogen storage with other energy assets. For example, the synergistic operation between hydrogen and electric vehicle (EV) fleets can enhance grid flexibility, as EVs can act as mobile storage units, supporting short-term demand response, while hydrogen provides deeper storage capabilities for extended grid balancing [197]. This layered approach to energy resource coordination strengthens grid resilience, particularly in networks with high renewable penetration.

ť		Applic	ations			Technical	l specs			Controller		Computational spe	cs
I	Frequency regulation	Voltage regulation	Grid balance	Grid congestion	Grid configuration	Conventional sources	RESs	ESTs	WEL/FC	I	Optimizer	Objective functions	Performance indices
8]	>	>	×	×	Two-area	•Deisel •Thermal	∙PV	×	>	IPD with one plus integral	Tunicate search algorithm	●ISE	Overshoot Undershoot Settling time
£	>	×	×	×	Hybrid power systems	•Deisel	•PV •WT	•BESS •FESS	>	Nonlinear PI	Dandelion optimizer	•IAE •ITAE •ISE	•Overshoot •Undershoot •Settling time
[5	>	×	×	×	Multi-micro grid	•Deisel •Thermal	•PV ●WT	•BESS	>	Fractional-order fuzzy PID	Modified harris hawks algorithm	•ITAE	OvershootUndershootBode plot
Ŧ	`	×	>	×	Hybrid power systems	•Deisel	•WT	•BESS	>	PID with filter	Genetic algorithm	●ITAE	 Overshoot Undershoot Settling time Peak time
5	>	×	×	×	Two-area	•Deisel •Thermal	₩•	•BESS •FESS	>	I, PI, PID	Smell agent Optimizer	●ISE	 Overshoot Undershoot Settling time
[5	>	×	>	×	Hybrid power systems	•Deisel	•PV •WT	•BESS •FESS	>	Id	Grey wolf optimizer	●ITAE	•Overshoot •Undershoot •Steady-state error
	>	×	×	×	Two-area	×	•PV •WT	•BESS •FESS	>	2-degree of freedom fuzzy PID	Modified multiverse algorithm	•IAE •ITAE •ISE	•Overshoot •Undershoot •Settling time
5	>	×	×	×	Isolated microgrid	•Deisel •Biogas	vq∙	•BESS	>	Cascade double- input interval Type 2 FLC	Enhanced salp swarm optimizer	•ITAE	S = = = = = = = = = = = = = = = = = = =
_	>	×	×	×	Isolated microgrid	•Deisel	•PV •WT	•BESS •FESS	>	Fractional order type-II fuzzy PID	Modified grey wolf optimizer	•ITAE	OvershootUndershootSettling time
_	>	×	×	×	Isolated microgrid	•Nuclear	×	•BESS	>	PID	Ant colony optimizer	•IAE •ITAE •ISE	 Overshoot Undershoot Settling time
	×	x	× ×	> ×	On-grid On-grid	× WN	MN MV	•BESS	>>	 × •Day-ahead model •Intra-day model 	× NN	 × Optimal operation schedule of WEL 	× ×
_	>	>	×	×	Shipboard microgrid	•Deisel	•PV ●WT	•BESS	>	Time-varying fractional order PID	Improved sine cosine algorithm	•ISE	OvershootUndershootSettling time

×	×	Isolated microgrid	×	•WT	•BESS	>	MPC	MN	•Frequency deviation	 Regulation
	×	Isolated microgrid	×	•PV •WT	•BESS	>	Intillegent MPC	Neural network	•MSE	 Peak PV and WT ramp
	×	On-grid	×	•WT	×	>	Dynamic frequency regulation	Analytical	×	 Frequency rate of change Nadir Steady-state
	>	On-grid	×	×	×	>	PI	Analytical	×	×
	×	Isolated	×	MN	×	>	MPC	ŇM	•Frequency	•Overshoot
	×	mıcrogrıd On-grid	×	×	×	>	PI	NM	deviation NM	settling timeOvershoot
										NadirSteady-state
	×	On-erid	×	Λq●	•SESS	>	Id	MN	MN	error ×
	×	Isolated	×	•WT	×	>	Fractional-order	Enhanced	●IAE	 Overshoot
		microgrid					PID	firefly optimizer		 settling time
	×	Two-area	Deisel	νq•	•BESS	>	PID	Artificial neural network	●ITAE	OvershootUndershoot
								algorithm		 Settling time

8. Economic perspectives of GHP

Fundamentally, conducting a thorough economic feasibility assessment is a crucial step in determining the viability and practicality of any project [210]. In fact, the cost of GHP in power systems is influenced by several factors, making it a complex and multi-faceted topic. The cost is primarily driven by the price of electricity, which accounts for a significant portion of the total production cost, often ranging between 50-70%. The WEL's efficiency is another critical factor, as higher efficiency reduces energy consumption and, consequently, the cost of hydrogen. WEL's capital cost, which includes the cost of the equipment, installation, and maintenance, also plays a significant role. WEL's type (e.g., PEM, AL, or SO) affects both efficiency and capital costs, with PEMWELs typically having higher costs but better adaptability to variable renewable energy inputs [211-213].

In addition, there are several external elements affecting the overall cost of GHP. These include government policies, subsidies, and carbon pricing, which can incentivize or penalize certain energy production methods. Technological advancements in WELs and RESs, as well as grid infrastructure improvements, can also reduce costs over time. Moreover, the cost of transporting and storing hydrogen, which can vary depending on the distance and method used, plays a substantial role in determining the final cost to the end-user. Ultimately, the integration of green hydrogen in power systems is expected to become more cost competitive as renewable energy prices decline, technology improves, and economies of scale are achieved [214, 215].

Numerically, the levelized cost of hydrogen (LCOH) is a metric used to evaluate the overall cost of producing hydrogen over the lifetime of a project. It's similar to the levelized cost of electricity (LCOE) and provides an average cost per kilogram of hydrogen produced, accounting for all expenses such as capital investment, operation, maintenance, and fuel costs, as formulated in (14) [216-218].

$$LCOH = \frac{C_{cex} + \sum_{t=1}^{n} (C_{om}^{t} + C_{elc}^{t}) \times (1+r)^{-t}}{\sum_{t=1}^{n} M_{H}^{t} \times (1+r)^{-t}}$$
(14)

where, the capital expenditure, initial investment cost for the WEL system, (\$) is represented by C_{cex} . C_{om}^{t} and C_{elc}^{t} are the operational and maintenance cost and electricity cost used for hydrogen production in year t, respectively. M_{H}^{t} , r, and t symbolize the amount of hydrogen produced in year t (kg), discount rate (the cost of capital or interest rate), and project lifetime (years), respectively. It's worth mentioning that load factor, stack degradation rates, and energy tariffs, especially if the WEL relies on intermittent renewables, are the key inputs for LCOH calculations.

In this context, several studies have focused on reducing the LCOH through the optimal design and configuration of WEL-based power systems. These efforts aim to enhance the economic viability of

GHP by optimizing system components, improving energy efficiency, and reducing overall costs. Table 9 summarizes some of the recent efforts in this regard.

Ref.	Grid	RESs	ESTs			WEL		H ₂ Tank	Location	LCOH
	mode			Туре	Capacity (kW)	Capacity factor (%)	Efficiency (%)	capacity (kg)		(\$/kg)
[219]	On	PV	BESS	PEM	900	NM	15.35 (Overall)	270	Canada	4.76*
[211]	Off	PV	BESS	PEM	NM	NM	90	10.97	Spain Italy	6.71- 7.82*
[212]	Off	PV	×	PEM	NM	90	61	×	NM	6.22
[213]	Off	PV	×	AL	4500	20	NM	NM	Korea	9.55
			BESS			22	Graphed			11.67
[214]	Off	WT	×	AL	4800	NM	80	×	Germany	4.84*
[215]	Off	WT	BESS	NM	NM	NM	75	3375	China	3.073- 3.155
[216]	On	PV	×	NM	30	10	<75	30	Turkey	1.78- 3.40
[217]	Off	WT	BESS	AL	250	NM	85	2022	NM	33.70
[218]	Off	WT	BESS	NM	750	NM	59	900	Australia	28.10
[220]	Off	PV	×	PEM	1000	NM	NM	×	Poland	14.13- 15.06*
[221]	Off	WT	×	PEM	185	NM	8.72 (Overall)	×	Badakhshan	3.887- 10.827
[222]	Off	PV/WT	×	AL	7.5	NM	62.8	NM	Egypt	4.54- 7.48
[223]	On	WT	×	PEM	3700-4400	60	85	11.45-12.75 (10 ³)	Croatia	17.1- 27.2
[224]	Off	PV/WT	BESS	PEM	250	NM	NM	700	Canada	21.9- 37.7
[225]	Off	PV/WT	BESS	NM	1500	NM	76.9	2000	India	3.00- 3.22
[226]	Off	PV	BESS	AL	70000	28	85	×	Australia	3.1
[227]	Off	WT	×	PEM	NM	NM	50.7	×	Algeria	6.1-6.8
[228]	On	PV	×	PEM/AL	62/49	NM	NM	×	Germany	6.83- 8.10
[229]	On	PV	×	PEM	800	NM	NM	1200	Oman	6.8
[230]	Off	Solar thermal /Geothermal	×	PEM	NM	NM	NM	NM	Iran	NM
[231]	Off	WT	×	AL PEM SO	NM	97 97 97	NM NM NM	NM	Chile	1.78- 2.45 2.61- 3.47 3.52- 4.11
[232]	Off	PV/WT	×	AL	7.5	NM	77	NM	Egypt	3.73- 4.656

Table 9. Analysis of published attempts to minimize hydrogen production costs

*Converted to \$ (browse the article for the original currency).

"Overall" denotes that the value corresponds to the total efficiency of the GHP system, accounting for all contributing sources.

Beyond LCOH, other financial indicators, such as net present value (NPV), internal rate of return (IRR), and payback period, offer a broader investment perspective [233]. Specifically, NPV determines whether the discounted sum of future cash flows is positive or negative, IRR identifies the discount rate at which NPV is zero, thus reflecting project profitability, and the payback period estimates the time required to recoup initial expenditures. When used in conjunction with LCOH, these metrics capture project-specific risk tolerance, making the analysis more robust for stakeholders and potential investors.

On the other side, given the uncertainties involved, sensitivity analyses are often performed to identify how changes in parameters, such as electricity price, WEL capital expenditure, efficiency, and policy incentives, impact GHP feasibility [213]. Scenario analyses can also explore various market conditions or policy frameworks, illuminating how GHP economics might evolve under different assumptions (e.g., high vs. moderate renewable deployment). These methodologies help pinpoint cost drivers and guide strategic planning, risk mitigation, or policy-making efforts [234].

9. Conclusion and future insights

In this article, a thorough review is presented to emphasize the transformative potential of green hydrogen as an EST through the integration of WELs into modern power grids. A key aspect highlighted in this work is the significance of the electrochemical model of WELs in understanding their internal dynamics, which enables accurate performance predictions, effective control design, and the optimization of hydrogen production processes. Furthermore, WELs have demonstrated their potential to enhance grid stability and reliability as a result of their diverse applications, from frequency and voltage control to grid congestion management and black start capabilities. It's worth highlighting that PEMWELs have the fastest dynamic response, enabling rapid adaptation to grid demand fluctuations, real-time supply balancing, and mitigating intermittent RES impacts. On the contrary, ALWELs boast lower capital costs, extended lifespans, and reduced degradation rates, making them an excellent choice for large-scale capacity projects. Moreover, the review underscores the importance of advanced control techniques, such as PI, FLC, FNN, and ANFIS, for optimizing DC converters connected to WELs/FCs. With ongoing research efforts focused on reducing hydrogen production costs, the future of green hydrogen in energy systems appears promising. This review not only sheds light on current technological advancements but also paves the way for future innovations in sustainable energy systems.

Herein, various future perspectives are presented to accelerate the shift towards GHP systems, aimed at enhancing power system stability and facilitating the large-scale penetration of RESs.

- Enhancing the precision of WEL mathematical models can lead to more accurate real-time operations and control strategies. Accordingly, novel mathematical models shall be developed to combine the whole operation aspects of GHP system, including hydrogen fluid dynamics in pipelines and storage medium.
- 2. Future advancements in WEL efficiency can significantly lower hydrogen production costs and enhance their role in power systems. For instance, developing more efficient and durable catalysts, such as using platinum-group metals or non-precious metal alternatives, can reduce energy losses and enhance reaction rates during electrolysis. GHP efficiency can also be enhanced by further

developments in PEM and AEM for augmenting ion conductivity. Moreover, better utilization of the generated heat during hydrogen production phase can improve the overall efficiency by directing it to preheat water or serve other thermal processes.

- 3. The integration between advanced converters topologies and control approaches can optimize power system stability. For example, the concept of grid-forming converters can provide ancillary services to power systems, like frequency and voltage regulation due to their rapid response to power fluctuations. Thus, they can optimally synchronize the hydrogen production with the grid's demands.
- 4. Recent evolution of AI-based controllers can further facilitate the utilization of GHP systems in dynamic grid support. Particularly, different machine learning control techniques, such as supervised learning (SL), unsupervised learning (UL), reinforcement learning (RL), and deep learning (DL), can significantly enhance the rapid integration of WELs. SL algorithms can predict energy demand patterns according to recorded historical data, while UL models determine the operation characteristics and abnormalities in WEL performance. Besides, RL techniques support dynamic operation of WELs in response to power oscillations by offering on-time decision-making capabilities. Lastly, DL approaches are appropriate for complex analyses of large datasets regarding GHP processes.

Conflicts of interest

There is no conflict of interest.

Data availability statement

The data that support the findings of this study are available from the corresponding author upon reasonable request.

Funders

No funding was received for this research.

References

[1] A. Talukdar, A. Chakrovorty, P. Sarmah, P. Paramasivam, V. Kumar, S.K. Yadav, S. Manickkam. A Review on Solid Oxide Fuel Cell Technology: An Efficient Energy Conversion System. International Journal of Energy Research. 2024 (2024) 6443247. <u>https://doi.org/10.1155/2024/6443247</u>.

[2] M. Sarwar, N.A. Warsi, A.S. Siddiqui, S. Kirmani. Optimal selection of renewable energy–based microgrid for sustainable energy supply. International Journal of Energy Research. 46 (2022) 5828-46. https://doi.org/10.1002/er.7525.

[3] S. Basnet, K. Deschinkel, L. Le Moyne, M.C. Péra. A review on recent standalone and grid integrated hybrid renewable energy systems: System optimization and energy management strategies. Renewable Energy Focus. 46 (2023) 103-25. <u>https://doi.org/10.1016/j.ref.2023.06.001</u>.

[4] S. Çelikdemir, M.T. Özdemir. A new approach in the cost estimation of a hydroelectric power plants in Türkiye based on geographical features. International Journal of Energy Research. 46 (2022) 20858-72. https://doi.org/10.1002/er.8384.

[5] A. Baba, D. Chandrasekharam. Geothermal resources for sustainable development: A case study. International Journal of Energy Research. 46 (2022) 20501-18. <u>https://doi.org/10.1002/er.7778</u>.

[6] H. Lee, B. Lee, M. Byun, H. Lim. Economic and environmental analysis for PEM water electrolysis based on replacement moment and renewable electricity resources. Energy Conversion and Management. 224 (2020) 113477. https://doi.org/10.1016/j.enconman.2020.113477.

[7] N. Nagano, R. Delage, T. Nakata. Optimal design and analysis of sector-coupled energy system in northeast Japan. Energies. 14 (2021) 2823. <u>https://doi.org/10.3390/en14102823</u>.

[8] B. Mbendane, B. Abe, C. Richards. Analysis of Supercapacitors in Renewable Energy Systems for Managing Power Fluctuations. International Journal of Energy Research. 2024 (2024) 4575761. https://doi.org/10.1155/2024/4575761.

[9] A. Mohammed, E.K. Sakr, M. Abo-Adma, R. Elazab. A comprehensive review of advancements and challenges in reactive power planning for microgrids. Energy Informatics. 7 (2024) 63. https://doi.org/10.1186/s42162-024-00341-3.

[10] D. Solyali, B. Safaei, O. Zargar, G. Aytac. A comprehensive state-of-the-art review of electrochemical battery storage systems for power grids. International Journal of Energy Research. 46 (2022) 17786-812. https://doi.org/10.1002/er.8451.

[11] A.W. Kumar, M.u.d. Mufti, M.Y. Zargar. Adaptive predictive control of flywheel storage for transient stability enhancement of a wind penetrated power system. International Journal of Energy Research. 46 (2022) 6654-71. <u>https://doi.org/10.1002/er.7600</u>.

[12] D. Wang, N. Liu, F. Chen, Y. Wang, J. Mao. Progress and prospects of energy storage technology research: Based on multidimensional comparison. Journal of Energy Storage. 75 (2024) 109710. https://doi.org/10.1016/j.est.2023.109710.

[13] M. Agabalaye-Rahvar, A. Mansour-Saatloo, M.A. Mirzaei, B. Mohammadi-Ivatloo, K. Zare. Economicenvironmental stochastic scheduling for hydrogen storage-based smart energy hub coordinated with integrated demand response program. International Journal of Energy Research. 45 (2021) 20232-57. https://doi.org/10.1002/er.7108.

[14] J. Lei, H. Ma, G. Qin, Z. Guo, P. Xia, C. Hao. A Comprehensive Review on the Power Supply System of Hydrogen Production Electrolyzers for Future Integrated Energy Systems. Energies. 17 (2024) 935. https://doi.org/10.3390/en17040935.

[15] K.T. Møller, T.R. Jensen, E. Akiba, H.-w. Li. Hydrogen-A sustainable energy carrier. Progress in natural science: Materials International. 27 (2017) 34-40. <u>https://doi.org/10.1016/j.pnsc.2016.12.014</u>.

[16] J. Wang, J. Wen, J. Wang, B. Yang, L. Jiang. Water electrolyzer operation scheduling for green hydrogen production: A review. Renewable and Sustainable Energy Reviews. 203 (2024) 114779. https://doi.org/10.1016/j.rser.2024.114779.

[17] B. Kanani, A. Zahedi. Step toward sustainable development through the integration of renewable energy systems with fuel cells: A review. Sustainable Energy Technologies and Assessments. 70 (2024) 103935. https://doi.org/10.1016/j.seta.2024.103935.

[18] S. Ramakrishnan, M. Delpisheh, C. Convery, D. Niblett, M. Vinothkannan, M. Mamlouk. Offshore green hydrogen production from wind energy: Critical review and perspective. Renewable and Sustainable Energy Reviews. 195 (2024) 114320. <u>https://doi.org/10.1016/j.rser.2024.114320</u>.

[19] S. Afroze, A.N.S.B. Sofri, M.S. Reza, Z.B. Iskakova, A. Kabyshev, K.A. Kuterbekov, K.Z. Bekmyrza, L. Taimuratova, M.R. Uddin, A.K. Azad. Solar-Powered Water Electrolysis Using Hybrid Solid Oxide Electrolyzer Cell (SOEC) for Green Hydrogen—A Review. Energies. 16 (2023) 7794. <u>https://doi.org/10.3390/en16237794</u>.

[20] D. Guilbert, S.M. Collura, A. Scipioni. DC/DC converter topologies for electrolyzers: State-of-the-art and remaining key issues. International Journal of Hydrogen Energy. 42 (2017) 23966-85. https://doi.org/10.1016/j.ijhydene.2017.07.174. [21] M. Koundi, H. El Fadil, Z. EL Idrissi, A. Lassioui, A. Intidam, T. Bouanou, S. Nady, A. Rachid. Investigation of hydrogen production system-based PEM EL: PEM EL modeling, DC/DC power converter, and controller design approaches. Clean Technologies. 5 (2023) 531-68. <u>https://doi.org/10.3390/cleantechnol5020028</u>.

[22] O. Vavilov, V. Yurkevich, D. Korobkov. Synthesis of an automatic control system for a voltage inverter as part of an autonomous power supply system with a hydrogen module. International Journal of Hydrogen Energy. 86 (2024) 742-50. https://doi.org/10.1016/j.ijhydene.2024.08.324.

[23] H. Rezk, A. Harrag. A robust type-2 fuzzy logic-based maximum power point tracking approach for thermoelectric generation systems. International Journal of Energy Research. 45 (2021) 18066-80. https://doi.org/10.1002/er.6955.

[24] Z. Cui, L. Wang, Q. Li, K. Wang. A comprehensive review on the state of charge estimation for lithium-ion battery based on neural network. International Journal of Energy Research. 46 (2022) 5423-40. https://doi.org/10.1002/er.7545.

[25] H. Rezk, A.G. Olabi, M.A. Abdelkareem, E.T. Sayed. Boosting the power density of two-chamber microbial fuel cell: Modeling and optimization. International Journal of Energy Research. 46 (2022) 20975-84. https://doi.org/10.1002/er.8589.

[26] A.J. Handique, R. Peer, J. Haas, J.C. Osorio-Aravena, L. Reyes-Chamorro. Distributed hydrogen systems: A literature review. International Journal of Hydrogen Energy. 85 (2024) 427-39. https://doi.org/10.1016/j.ijhydene.2024.08.206.

[27] J. Li, G. Fu, X. Sheng, G. Li, H. Chen, K. Shu, Y. Dong, T. Wang, Y. Deng. A Comprehensive Review on Catalysts for Seawater Electrolysis. Advanced Powder Materials. (2024) 100227. https://doi.org/10.1016/j.apmate.2024.100227.

[28] F.P. Martins, S.D.-L. Almaraz, A.B.B. Junior, C. Azzaro-Pantel, P. Parikh. Hydrogen and the sustainable development goals: Synergies and trade-offs. Renewable and Sustainable Energy Reviews. 204 (2024) 114796. https://doi.org/10.1016/j.rser.2024.114796.

[29] R. Cozzolino, G. Bella. A review of electrolyzer-based systems providing grid ancillary services: current status, market, challenges and future directions. Frontiers in Energy Research. 12 (2024) 1358333. https://doi.org/10.3389/fenrg.2024.1358333.

[30] M. Awad, A. Said, M.H. Saad, A. Farouk, M.M. Mahmoud, M.S. Alshammari, M.L. Alghaythi, S.H.A. Aleem, A.Y. Abdelaziz, A.I. Omar. A review of water electrolysis for green hydrogen generation considering PV/wind/hybrid/hydropower/geothermal/tidal and wave/biogas energy systems, economic analysis, and its application. Alexandria Engineering Journal. 87 (2024) 213-39. <u>https://doi.org/10.1016/j.aej.2023.12.032</u>.

[31] M.B. Hossain, M.R. Islam, K.M. Muttaqi, D. Sutanto, A.P. Agalgaonkar. Advancement of fuel cells and electrolyzers technologies and their applications to renewable-rich power grids. Journal of Energy Storage. 62 (2023) 106842. <u>https://doi.org/10.1016/j.est.2023.106842</u>.

[32] M.A. Hossain, M.R. Islam, M.A. Hossain, M. Hossain. Control strategy review for hydrogen-renewable energy power system. Journal of Energy Storage. 72 (2023) 108170. https://doi.org/10.1016/j.est.2023.108170.

[33] P. Vaghela, V. Pandey, A. Sircar, K. Yadav, N. Bist, R. Kumari. Energy storage techniques, applications, and recent trends: A sustainable solution for power storage. MRS Energy & Sustainability. 10 (2023) 261-76. https://doi.org/10.1557/s43581-023-00069-9.

[34] R.A. Abdelsalam, M. Mohamed, H.E. Farag, E.F. El-Saadany. Green hydrogen production plants: A technoeconomic review. Energy Conversion and Management. 319 (2024) 118907. https://doi.org/10.1016/j.enconman.2024.118907.

[35] N. Thakkar, P. Paliwal. Hydrogen storage based micro-grid: A comprehensive review on technology, energy management and planning techniques. International Journal of Green Energy. 20 (2023) 445-63. https://doi.org/10.1080/15435075.2022.2049797.

[36] N. Lystbæk, M. Gregersen, H.R. Shaker. Review of Energy Portfolio Optimization in Energy Markets Considering Flexibility of Power-to-X. Sustainability. 15 (2023) 4422. <u>https://doi.org/10.3390/su15054422</u>.

[37] A. R. Singh, R.S. Kumar, M. Bajaj, C.B. Khadse, I. Zaitsev. Machine learning-based energy management and power forecasting in grid-connected microgrids with multiple distributed energy sources. Scientific Reports. 14 (2024) 19207. <u>https://doi.org/10.1038/s41598-024-70336-3</u>.

[38] P.K. Bera, S.R. Pani, C. Isik, R.C. Bansal. A hybrid intelligent system for protection of transmission lines connected to PV farms based on linear trends. Electric Power Systems Research. 237 (2024) 110991. https://doi.org/10.1016/j.epsr.2024.110991.

[39] I.F. Davoudkhani, P. Zare, A.Y. Abdelaziz, M. Bajaj, M.B. Tuka. Robust load-frequency control of islanded urban microgrid using 1PD-3DOF-PID controller including mobile EV energy storage. Scientific Reports. 14 (2024) 13962. <u>https://doi.org/10.1038/s41598-024-64794-y</u>.

[40] S.H. Rouhani, C.-L. Su, S. Mobayen, N. Razmjooy, M. Elsisi. Cyber resilience in renewable Microgrids: A Review of Standards, Challenges, and Solutions. Energy. (2024) 133081. https://doi.org/10.1016/j.energy.2024.133081.

[41] S. Behzadi, A. Bagheri, A. Rabiee. Resilience-Oriented Operation of Micro-Grids in both Grid-Connected and Isolated Conditions within Sustainable Active Distribution Networks. arXiv preprint arXiv:240319147. (2024). https://doi.org/10.22098/joape.2023.12142.1902.

[42] W. Bai, Z. Zhang, Y. Zhang, W. Zhao, B. Zhang. Virtual coupling control of photovoltaic-energy storage power generation system for efficient power support. Energy Reports. 12 (2024) 1742-52. https://doi.org/10.1016/j.egyr.2024.07.052.

[43] H.O. Shami, A. Basem, A.H. Al-Rubaye, K. Sabzevari. A novel strategy to enhance power management in AC/DC hybrid microgrid using virtual synchronous generator based interlinking converters integrated with energy storage system. Energy Reports. 12 (2024) 75-94. <u>https://doi.org/10.1016/j.egyr.2024.06.001</u>.

[44] D.A. Elalfy, E. Gouda, M.F. Kotb, V. Bureš, B.E. Sedhom. Comprehensive review of energy storage systems technologies, objectives, challenges, and future trends. Energy Strategy Reviews. 54 (2024) 101482. https://doi.org/10.1016/j.esr.2024.101482.

[45] Z. Zhang, T. Ding, Q. Zhou, Y. Sun, M. Qu, Z. Zeng, Y. Ju, L. Li, K. Wang, F. Chi. A review of technologies and applications on versatile energy storage systems. Renewable and Sustainable Energy Reviews. 148 (2021) 111263. <u>https://doi.org/10.1016/j.rser.2021.111263</u>.

[46] F. Nadeem, S.S. Hussain, P.K. Tiwari, A.K. Goswami, T.S. Ustun. Comparative review of energy storage systems, their roles, and impacts on future power systems. IEEE access. 7 (2018) 4555-85. https://doi.org/10.1109/ACCESS.2018.2888497.

[47] M.S. Nazir, A.N. Abdalla, A.S. M. Metwally, M. Imran, P. Bocchetta, M.S. Javed. Cryogenic-energy-storage-based optimized green growth of an integrated and sustainable energy system. Sustainability. 14 (2022) 5301.
[48] Z. Ziobrowski, A. Rotkegel. Assessment of Hydrogen Energy Industry Chain Based on Hydrogen Production Methods, Storage, and Utilization. Energies. 17 (2024) 1808. <u>https://doi.org/10.3390/en17081808</u>.

[49] F. Dawood, M. Anda, G. Shafiullah. Hydrogen production for energy: An overview. International Journal of Hydrogen Energy. 45 (2020) 3847-69. <u>https://doi.org/10.1016/j.ijhydene.2019.12.059</u>.

[50] P. Saha, F.A. Akash, S.M. Shovon, M.U. Monir, M.T. Ahmed, M.F.H. Khan, S.M. Sarkar, M.K. Islam, M.M. Hasan, D.-V.N. Vo. Grey, blue, and green hydrogen: A comprehensive review of production methods and prospects for zero-emission energy. International Journal of Green Energy. 21 (2024) 1383-97. https://doi.org/10.1080/15435075.2023.2244583.

[51] R. Li, H. Kawanami. A recent review of primary hydrogen carriers, hydrogen production methods, and applications. Catalysts. 13 (2023) 562. <u>https://doi.org/10.3390/catal13030562</u>.

[52] S.K. Dash, S. Chakraborty, D. Elangovan. A brief review of hydrogen production methods and their challenges. Energies. 16 (2023) 1141. <u>https://doi.org/10.3390/en16031141</u>.

[53] M. Fochesato, C. Peter, L. Morandi, J. Lygeros. Peak shaving with hydrogen energy storage: From stochastic control to experiments on a 4 MWh facility. Applied Energy. 376 (2024) 123965. https://doi.org/10.1016/j.apenergy.2024.123965.

[54] P. Patel, V. Garaniya, T. Baalisampang, E. Arzaghi, R. Abbassi, F. Salehi. A technical review on quantitative risk analysis for hydrogen infrastructure. Journal of Loss Prevention in the Process Industries. (2024) 105403. https://doi.org/10.1016/j.jlp.2024.105403.

[55] A. Szyszka. Ten years of solar hydrogen demonstration project at Neunburg vorm Wald, Germany. International Journal of Hydrogen Energy. 23 (1998) 849-60. <u>https://doi.org/10.1016/S0360-3199(97)00172-9</u>.

[56] C. Meurer, H. Barthels, W. Brocke, B. Emonts, H. Groehn. PHOEBUS—an autonomous supply system with renewable energy: six years of operational experience and advanced concepts. Solar energy. 67 (1999) 131-8. https://doi.org/10.1016/S0038-092X(00)00043-8.

[57] A. Yilanci, I. Dincer, H.K. Ozturk. A review on solar-hydrogen/fuel cell hybrid energy systems for stationary applications. Progress in energy and combustion science. 35 (2009) 231-44. https://doi.org/10.1016/j.pecs.2008.07.004.

[58] K. Voss, A. Goetzberger, G. Bopp, A. Häberle, A. Heinzel, H. Lehmberg. The self-sufficient solar house in Freiburg—Results of 3 years of operation. Solar Energy. 58 (1996) 17-23. <u>https://doi.org/10.1016/0038-092X(96)00046-1</u>.

[59] H. Barthels, W. Brocke, K. Bonhoff. An autonomous energy supply system comprising photovoltaics, electrolytic hydrogen, fuel cell. Proceedings of the 11th World Hydrogen Energy Conference, Stuttgart1996. pp. 23-8. doi:<u>https://doi.org/10.1016/S0360-3199(97)00055-4</u>.

[60] K. Agbossou, M.L. Kolhe, J. Hamelin, É. Bernier, T.K. Bose. Electrolytic hydrogen based renewable energy system with oxygen recovery and re-utilization. Renewable energy. 29 (2004) 1305-18. https://doi.org/10.1016/j.renene.2003.12.006.

[61] S. Kélouwani, K. Agbossou, R. Chahine. Model for energy conversion in renewable energy system with hydrogen storage. Journal of Power Sources. 140 (2005) 392-9. https://doi.org/10.1016/j.jpowsour.2004.08.019.

[62] A. Chaparro, J. Soler, M. Escudero, E. De Ceballos, U. Wittstadt, L. Daza. Data results and operational experience with a solar hydrogen system. Journal of power sources. 144 (2005) 165-9. https://doi.org/10.1016/j.jpowsour.2004.12.044.

[63] Ø. Ulleberg, T. Nakken, A. Ete. The wind/hydrogen demonstration system at Utsira in Norway: Evaluation of system performance using operational data and updated hydrogen energy system modeling tools. International journal of hydrogen energy. 35 (2010) 1841-52. <u>https://doi.org/10.1016/j.ijhydene.2009.10.077</u>.
[64] Z. Chehade, C. Mansilla, P. Lucchese, S. Hilliard, J. Proost. Review and analysis of demonstration projects on power-to-X pathways in the world. International Journal of Hydrogen Energy. 44 (2019) 27637-55. <u>https://doi.org/10.1016/j.ijhydene.2019.08.260</u>.

[65] M. Thema, F. Bauer, M. Sterner. Power-to-Gas: Electrolysis and methanation status review. Renewable and Sustainable Energy Reviews. 112 (2019) 775-87. <u>https://doi.org/10.1016/j.rser.2019.06.030</u>.

[66] E. Cetin, A. Yilanci, Y. Oner, M. Colak, I. Kasikci, H.K. Ozturk. Electrical analysis of a hybrid photovoltaichydrogen/fuel cell energy system in Denizli, Turkey. Energy and Buildings. 41 (2009) 975-81. https://doi.org/10.1016/j.enbuild.2009.04.004.

[67] C. Wulf, J. Linßen, P. Zapp. Review of power-to-gas projects in Europe. Energy Procedia. 155 (2018) 367-78. <u>https://doi.org/10.1016/j.egypro.2018.11.041</u>.

[68] M. Sartory, E. Wallnöfer-Ogris, P. Salman, T. Fellinger, M. Justl, A. Trattner, M. Klell. Theoretical and experimental analysis of an asymmetric high pressure PEM water electrolyser up to 155 bar. International Journal of Hydrogen Energy. 42 (2017) 30493-508. <u>https://doi.org/10.1016/j.ijhydene.2017.10.112</u>.

[69] M. Persson, D. Mignard, D. Hogg. Data Averaging for Modelling Energy Storage: What Information is Hidden? , 2018 53rd International Universities Power Engineering Conference (UPEC). IEEE2018. pp. 1-5. doi:<u>https://doi.org/10.1109/UPEC.2018.8541917</u>.

[70] T. Isaac. HyDeploy: The UK's first hydrogen blending deployment project. Clean Energy. 3 (2019) 114-25. https://doi.org/10.1093/ce/zkz006.

[71] M.A. Gallo, J.G.G. Clúa. Sizing and analytical optimization of an alkaline water electrolyzer powered by a grid-assisted wind turbine to minimize grid power exchange. Renewable Energy. 216 (2023) 118990. https://doi.org/10.1016/j.renene.2023.118990.

[72] Y. Chen, J. Chen, K. Bai, J. Liu, Z. Xiao, S. Fan. Energy consumption model of flow-through electrolyzer considering mass transfer overpotential for hydrogen production from alkaline water splitting. Journal of Power Sources. 621 (2024) 235292. <u>https://doi.org/10.1016/j.jpowsour.2024.235292</u>.

[73] A. Meriläinen, A. Kosonen, J. Jokisalo, R. Kosonen, P. Kauranen, J. Ahola. Techno-economic evaluation of waste heat recovery from an off-grid alkaline water electrolyzer plant and its application in a district heating network in Finland. Energy. 306 (2024) 132181. <u>https://doi.org/10.1016/j.energy.2024.132181</u>.

[74] Q. Xu, L. Zhang, J. Zhang, J. Wang, Y. Hu, H. Jiang, C. Li. Anion exchange membrane water electrolyzer: electrode design, lab-scaled testing system and performance evaluation. EnergyChem. 4 (2022) 100087. https://doi.org/10.1016/j.enchem.2022.100087.

[75] Y. Chen, C. Liu, J. Xu, C. Xia, P. Wang, B.Y. Xia, Y. Yan, X. Wang. Key components and design strategy for a proton exchange membrane water electrolyzer. Small Structures. 4 (2023) 2200130. https://doi.org/10.1002/sstr.202200130.

[76] B. Łosiewicz. Technology for Green Hydrogen Production: Desk Analysis. Energies. 17 (2024) 4514. https://doi.org/10.3390/en17174514.

[77] S.S. Kumar, H. Lim. An overview of water electrolysis technologies for green hydrogen production. Energy reports. 8 (2022) 13793-813. <u>https://doi.org/10.1016/j.egyr.2022.10.127</u>.

[78] M.K. Ratib, K.M. Muttaqi, M.R. Islam, D. Sutanto, A.P. Agalgaonkar. Electrical circuit modeling of proton exchange membrane electrolyzer: The state-of-the-art, current challenges, and recommendations. International Journal of Hydrogen Energy. (2023). <u>https://doi.org/10.1016/j.ijhydene.2023.08.319</u>.

[79] M. He, G. Nie, H. Yang, B. Li, S. Zhou, X. Wang, X. Meng. A generic equivalent circuit model for PEM electrolyzer with multi-timescale and stages under multi-mode control. Applied Energy. 359 (2024) 122728. https://doi.org/10.1016/j.apenergy.2024.122728.

[80] F. Gambou, D. Guilbert, M. Zasadzinski, H. Rafaralahy. A comprehensive survey of alkaline electrolyzer modeling: electrical domain and specific electrolyte conductivity. Energies. 15 (2022) 3452. https://doi.org/10.3390/en15093452.

[81] D. Hidouri, R. Marouani, A. Cherif. Modeling and Simulation of a Renewable Energy PV/PEM with Green Hydrogen Storage. Engineering, Technology & Applied Science Research. 14 (2024) 12543-8. https://doi.org/10.48084/etasr.6492.

[82] A. Elmaihy, M. Amin, M. Bennaya, A. Rashad. Thermodynamic modeling of alkaline water electrolyzer and assessment of reported cell voltages correlations at low temperature and atmospheric pressure: Critical review. Journal of Energy Storage. 96 (2024) 112674. <u>https://doi.org/10.1016/j.est.2024.112674</u>.

[83] D. Falcão, A. Pinto. A review on PEM electrolyzer modelling: Guidelines for beginners. Journal of cleaner production. 261 (2020) 121184. <u>https://doi.org/10.1016/j.jclepro.2020.121184</u>.

[84] A. Majumdar, M. Haas, I. Elliot, S. Nazari. Control and control-oriented modeling of PEM water electrolyzers: A review. International Journal of Hydrogen Energy. 48 (2023) 30621-41. https://doi.org/10.1016/j.ijhydene.2023.04.204.

[85] A.I. Dawahdeh, R.A. Al-Shdeifat, A.-N. Moh'd A. Power-to-X system utilizing an advanced solar system integrated with a thermally regenerative electrochemical cycle. Energy. (2024) 131959. https://doi.org/10.1016/j.rser.2019.06.030.

[86] X. Mao, Y. Tian, A. Yang, G. Zhang. Identification of Equivalent Circuit Parameters for Proton Exchange Membrane Electrolyzer (PEM) Engineering Models. IEE Access. (2024). https://doi.org/10.1109/ACCESS.2024.3358456.

[87] A.M. Abomazid, N.A. El-Taweel, H.E. Farag. Novel analytical approach for parameters identification of PEM electrolyzer. IEEE Transactions on Industrial Informatics. 18 (2021) 5870-81. https://doi.org/10.1109/TII.2021.3132941.

[88] R. Khajuria, R. Lamba, R. Kumar. Optimal parameter identification of pem electrolyzer using bald eagle search optimization algorithm. 2022 IEEE 10th Power India International Conference (PIICON). IEEE2022. pp. 1-6. doi:<u>https://doi.org/10.1109/PIICON56320.2022.10045282</u>.

[89] Á. Iribarren, D. Elizondo, E.L. Barrios, H. Ibaiondo, A. Sanchez-Ruiz, J. Arza, P. Sanchis, A. Ursúa. Dynamic modeling of a pressurized alkaline water electrolyzer: a multiphysics approach. IEEE Transactions on Industry Applications. 59 (2023) 3741-53. <u>https://doi.org/10.1109/TIA.2023.3247405</u>.

[90] A.G. Vidales, N.C. Millan, C. Bock. Modeling of anion exchange membrane water electrolyzers: The influence of operating parameters. Chemical Engineering Research and Design. 194 (2023) 636-48. https://doi.org/10.1016/j.cherd.2023.05.004.

[91] J. Koh, D. Yoon, C.H. Oh. Simple electrolyzer model development for high-temperature electrolysis system analysis using solid oxide electrolysis cell. Journal of Nuclear Science and technology. 47 (2010) 599-607. https://doi.org/10.1080/18811248.2010.9720957. [92] M. Dreoni, F. Balduzzi, F. Ferro, G. Fagioli, K. Panichi, G. Ferrara, A. Bianchini. Multi-physics model development and application to real-case alkaline electrolyzer. Journal of Physics: Conference Series. IOP Publishing2023. p. 012060. doi:https://doi.org/10.1088/1742-6596/2648/1/012060.

[93] M. Carmo, D.L. Fritz, J. Mergel, D. Stolten. A comprehensive review on PEM water electrolysis. International journal of hydrogen energy. 38 (2013) 4901-34. <u>https://doi.org/10.1016/j.ijhydene.2013.01.151</u>.
[94] F. Barbir. PEM electrolysis for production of hydrogen from renewable energy sources. Solar energy. 78 (2005) 661-9. <u>https://doi.org/10.1016/j.solener.2004.09.003</u>.

[95] S.-C. Chang, R.-E. Gu, Y.-H. Chan. Parameter Analysis of Anion Exchange Membrane Water Electrolysis System by Numerical Simulation. Energies. 17 (2024) 5682. <u>https://doi.org/10.3390/en17225682</u>.

[96] V. Menon, V.M. Janardhanan, O. Deutschmann. A mathematical model to analyze solid oxide electrolyzer cells (SOECs) for hydrogen production. Chemical Engineering Science. 110 (2014) 83-93. https://doi.org/10.1016/j.ces.2013.10.025.

[97] B. Yodwong, D. Guilbert, M. Phattanasak, W. Kaewmanee, M. Hinaje, G. Vitale. AC-DC converters for electrolyzer applications: State of the art and future challenges. Electronics. 9 (2020) 912. https://doi.org/10.3390/electronics9060912.

[98] F. Alonge, S.M. Collura, F. D'ippolito, D. Guilbert, M. Luna, G. Vitale. Design of a robust controller for DC/DC converter–electrolyzer systems supplied by μ WECSs subject to highly fluctuating wind speed. Control Engineering Practice. 98 (2020) 104383. <u>https://doi.org/10.1016/j.conengprac.2020.104383</u>.

[99] M.C. Mira, Z. Zhang, K.L. Jørgensen, M.A. Andersen. Fractional charging converter with high efficiency and low cost for electrochemical energy storage devices. IEEE Transactions on Industry Applications. 55 (2019) 7461-70. <u>https://doi.org/10.1109/TIA.2019.2921295</u>.

[100] M. Annamalai. A comprehensive review on isolated and non-isolated converter configuration and fast charging technology: For battery and plug in hybrid electric vehicle. Heliyon. (2023). https://doi.org/10.1016/j.heliyon.2023.e18808.

[101] F.L. Tofoli, T.M.J. Carlos, A.S. Morais. Review, Properties, and Synthesis of Single-Switch Non-Isolated DC-DC Converters with a Wide Conversion Range. Sensors. 24 (2024) 2264. <u>https://doi.org/10.3390/s24072264</u>.

[102] S. Bairabathina, S. Balamurugan. Review on non-isolated multi-input step-up converters for gridindependent hybrid electric vehicles. International journal of hydrogen energy. 45 (2020) 21687-713. https://doi.org/10.1016/j.ijhydene.2020.05.277.

[103] F.A. Abbas, T.A. Abdul-Jabbar, A.A. Obed, A. Kersten, M. Kuder, T. Weyh. A comprehensive review and analytical comparison of non-isolated dc-dc converters for fuel cell applications. Energies. 16 (2023) 3493. https://doi.org/10.3390/en16083493.

[104] P. Cervellini, M. Carnaghi, P. Antoszczuk, R.G. Retegui, M. Funes. Input and output total currents characterization in BCM and CCM Interleaved Power Converters Under Inductance Mismatch. Power Electronic Devices and Components. 3 (2022) 100016. <u>https://doi.org/10.1016/j.pedc.2022.100016</u>.

[105] T. Sutikno, R.A. Aprilianto, H.S. Purnama. Application of non-isolated bidirectional DC–DC converters for renewable and sustainable energy systems: a review. Clean Energy. 7 (2023) 293-311. https://doi.org/10.1093/ce/zkac070.

[106] V. Samavatian, A. Radan. A novel low-ripple interleaved buck–boost converter with high efficiency and low oscillation for fuel-cell applications. International Journal of Electrical Power & Energy Systems. 63 (2014) 446-54. <u>https://doi.org/10.1016/j.ijepes.2014.06.020</u>.

[107] M.E. Şahin, H.İ. Okumuş, M.T. Aydemir. Implementation of an electrolysis system with DC/DC synchronous buck converter. International Journal of hydrogen energy. 39 (2014) 6802-12. https://doi.org/10.1016/j.ijhydene.2014.02.084.

[108] E. Carbajal-Gutierrez, J. Morales-Saldana, J. Leyva-Ramos. Modeling of a single-switch quadratic buck converter. IEEE transactions on aerospace and electronic systems. 41 (2005) 1450-6. https://doi.org/10.1109/TAES.2005.1561895.

[109] F.L. de Sá, C.V. Eiterer, D. Ruiz-Caballero, S.A. Mussa. Double quadratic buck converter. 2013 Brazilian Power Electronics Conference. IEEE2013. pp. 36-43. doi:<u>https://doi.org/10.1109/COBEP.2013.6785092</u>.

[110] S. Birca-Galateanu. Triple step-down DC-DC converters. PESC Record 27th Annual IEEE Power Electronics Specialists Conference. IEEE1996. pp. 408-13. doi:<u>https://doi.org/10.1109/PESC.1996.548612</u>.

[111] S. Xiong, S.-C. Wong, S.-C. Tan, K.T. Chi. A family of exponential step-down switched-capacitor converters and their applications in two-stage converters. IEEE Transactions on Power Electronics. 29 (2013) 1870-80. https://doi.org/10.1109/TPEL.2013.2270290.

[112] D. Guilbert, A. N'Diaye, A. Gaillard, A. Djerdir. Fuel cell systems reliability and availability enhancement by developing a fast and efficient power switch open-circuit fault detection algorithm in interleaved DC/DC boost converter topologies. International Journal of Hydrogen Energy. 41 (2016) 15505-17. https://doi.org/10.1016/j.ijhydene.2016.01.169.

[113] P. Thounthong, B. Davat. Study of a multiphase interleaved step-up converter for fuel cell high power applications. Energy Conversion and Management. 51 (2010) 826-32. https://doi.org/10.1016/j.enconman.2009.11.018.

[114] V.F. Pires, E. Romero-Cadaval, D. Vinnikov, I. Roasto, J. Martins. Power converter interfaces for electrochemical energy storage systems—A review. Energy conversion and management. 86 (2014) 453-75. https://doi.org/10.1016/j.enconman.2014.05.003.

[115] A. Kirubakaran, S. Jain, R. Nema. A review on fuel cell technologies and power electronic interface. Renewable and sustainable energy reviews. 13 (2009) 2430-40. <u>https://doi.org/10.1016/j.rser.2009.04.004</u>.

[116] C. Charin, S. Iqbal, S. Taib. A New Soft Switching PWM DC-DC Converter with Auxiliary Circuit and Centre-Tapped Transformer Rectifier. Procedia Engineering. 53 (2013) 241-7. https://doi.org/10.1016/j.proeng.2013.02.032.

[117] R. Pittini, Z. Zhang, M.A. Andersen. Isolated full bridge boost DC-DC converter designed for bidirectional operation of fuel cells/electrolyzer cells in grid-tie applications. 2013 15th European Conference on Power Electronics and Applications (EPE). IEEE2013. pp. 1-10. doi:https://doi.org/10.1109/EPE.2013.6634433.

[118] Z. Xuan, H. Shenghua, N. Guoyun. A three-phase dual active bridge bidirectional zvs dc/dc converter. Physics Procedia. 24 (2012) 139-48. <u>https://doi.org/10.1016/j.phpro.2012.02.022</u>.

[119] M. Alaraj, Z.J. Ren, J.-D. Park. Microbial fuel cell energy harvesting using synchronous flyback converter. Journal of Power Sources. 247 (2014) 636-42. <u>https://doi.org/10.1016/j.jpowsour.2013.09.017</u>.

[120] R. Gopinath, S. Kim, J.-H. Hahn, P.N. Enjeti, M.B. Yeary, J.W. Howze. Development of a low cost fuel cell inverter system with DSP control. IEEE transactions on Power Electronics. 19 (2004) 1256-62. https://doi.org/10.1109/TPEL.2004.833432.

[121] R. Pittini, Z. Zhang, M.A. Andersen. Analysis of DC/DC converter efficiency for energy storage system based on bidirectional fuel cells. IEEE PES ISGT Europe 2013. IEEE2013. pp. 1-5. doi:<u>https://doi.org/10.1109/ISGTEurope.2013.6695344</u>.

[122] M. Banaei, E. Salary. Application of multi-stage converter in distributed generation systems. Energy conversion and management. 62 (2012) 76-83. <u>https://doi.org/10.1016/j.enconman.2012.03.026</u>.

[123] A. Blinov, A. Andrijanovits. New DC/DC converter for electrolyser interfacing with stand-alone renewable energy system. Electrical, Control and Communication Engineering. 1 (2012) 24-9. https://doi.org/10.2478/v10314-012-0004-1.

[124] D.S. Gautam, A.K. Bhat. A comparison of soft-switched DC-to-DC converters for electrolyzer application. IEEE Transactions on power electronics. 28 (2012) 54-63. https://doi.org/10.1109/TPEL.2012.2195682.

[125] A. Veerendra, K. Peddakapu. Enhancement of a grid-connected DFIG wind turbine system using fractional order PI controllers. Renewable Energy Focus. 47 (2023) 100506. <u>https://doi.org/10.1016/j.ref.2023.100506</u>.

[126] S. Mohiuddin, M.A. Mahmud, A. Haruni, H. Pota. Design and implementation of partial feedback linearizing controller for grid-connected fuel cell systems. International Journal of Electrical Power & Energy Systems. 93 (2017) 414-25. <u>https://doi.org/10.1016/j.ijepes.2017.06.015</u>.

[127] H.R. Esmaeilian, R. Fadaeinedjad. Resolving power quality issues raised by aerodynamic aspects of wind turbine in isolated microgrids using fuel cell/electrolyzer system. IEEE transactions on sustainable energy. 7 (2016) 1274-83. <u>https://doi.org/10.1109/TSTE.2016.2544102</u>.

[128] M.G. Dozein, A.M. De Corato, P. Mancarella. Virtual inertia response and frequency control ancillary services from hydrogen electrolyzers. IEEE Transactions on Power Systems. 38 (2022) 2447-59. https://doi.org/10.1109/TPWRS.2022.3181853.

[129] A.S. Zinober. An introduction to sliding mode variable structure control. Variable structure and Lyapunov control2005. pp. 1-22. <u>https://doi.org/10.1007/BFb0033676</u>.

[130] V. Utkin, A. Poznyak, Y. Orlov, A. Polyakov. Conventional and high order sliding mode control. Journal of the Franklin Institute. 357 (2020) 10244-61. <u>https://doi.org/10.1016/j.jfranklin.2020.06.018</u>.

[131] I. Sami, S. Ullah, L. Khan, A. Al-Durra, J.-S. Ro. Integer and fractional-order sliding mode control schemes in wind energy conversion systems: Comprehensive review, comparison, and technical insight. Fractal and Fractional. 6 (2022) 447. <u>https://doi.org/10.3390/fractalfract6080447</u>.

[132] F. Akter, T.K. Roy, M.S. Islam, A.F. Alkhateeb, M.A. Mollah. Design of a nonlinear integral terminal sliding mode controller for a pem fuel cell based on a dc-dc boost converter. IEEE Access. 10 (2022) 97419-28. https://doi.org/10.1109/ACCESS.2022.3205733.

[133] W. Wang, C. Wen, J. Zhou. Adaptive backstepping control of uncertain systems with actuator failures, subsystem interactions, and nonsmooth nonlinearities. CRC Press2017.

[134] X. Li, X. Fang. Passive backstepping control of dual active bridge converter in modular three-port DC converter. Electronics. 12 (2023) 1074. <u>https://doi.org/10.3390/electronics12051074</u>.

[135] X. Hao, I. Salhi, S. Laghrouche, Y. Ait-Amirat, A. Djerdir. Backstepping supertwisting control of four-phase interleaved boost converter for PEM fuel cell. IEEE Transactions on Power Electronics. 37 (2022) 7858-70. https://doi.org/10.1109/TPEL.2022.3149099.

[136] W.R. Sultana, S.K. Sahoo, S. Sukchai, S. Yamuna, D. Venkatesh. A review on state of art development of model predictive control for renewable energy applications. Renewable and sustainable energy reviews. 76 (2017) 391-406. <u>https://doi.org/10.1016/j.rser.2017.03.058</u>.

[137] E. Irmak, N. Güler. A model predictive control-based hybrid MPPT method for boost converters. International Journal of Electronics. 107 (2020) 1-16. <u>https://doi.org/10.1080/00207217.2019.1582715</u>.

[138] A. hamlat, M.h. Sekour, M. mankour, M. yahiaoui, M. khalfaoui, B. brahmi. Advanced Power Management and Control Using Fuzzy Backstepping Super-Twisting Controls Designed for Fuel Cell Supercapacitors Hybrid Power Systems for Traction Applications. Journal of Control, Automation and Electrical Systems. 34 (2023) 996-1012. <u>https://doi.org/10.1007/s40313-023-01014-4</u>.

[139] S. Vazquez, J.I. Leon, L.G. Franquelo, J. Rodriguez, H.A. Young, A. Marquez, P. Zanchetta. Model predictive control: A review of its applications in power electronics. IEEE industrial electronics magazine. 8 (2014) 16-31. https://doi.org/10.1109/MIE.2013.2290138.

[140] A. Beşkardeş, Y. Hameş, K. Kaya. A comprehensive review on fuzzy logic control systems for all, hybrid, and fuel cell electric vehicles. Soft Computing. (2024) 1-39. <u>https://doi.org/10.1007/s00500-023-09454-5</u>.

[141] J. Du, X. Zhao, X. Liu, G. Liu, Y. Xiong. A Fuzzy Logic Control-Based Approach for Real-Time Energy Management of the Fuel Cell Electrical Bus Considering the Durability of the Fuel Cell System. World Electric Vehicle Journal. 15 (2024) 92. <u>https://doi.org/10.3390/wevj15030092</u>.

[142] T. Chaiyatham, I. Ngamroo. A bee colony optimization based-fuzzy logic-PID control design of electrolyzer for microgrid stabilization. Int J Innov Comput Inf Control. 8 (2012) 6049-66.

[143] D. Svozil, V. Kvasnicka, J. Pospichal. Introduction to multi-layer feed-forward neural networks. Chemometrics and intelligent laboratory systems. 39 (1997) 43-62. <u>https://doi.org/10.1016/S0169-7439(97)00061-0</u>.

[144] A. Harrag, H. Bahri. Novel neural network IC-based variable step size fuel cell MPPT controller: performance, efficiency and lifetime improvement. International journal of hydrogen energy. 42 (2017) 3549-63. <u>https://doi.org/10.1016/j.ijhydene.2016.12.079</u>.

[145] M.I. Guerra, F.M. de Araújo, J.T. de Carvalho Neto, R.G. Vieira. Survey on adaptative neural fuzzy inference system (ANFIS) architecture applied to photovoltaic systems. Energy Systems. 15 (2024) 505-41. https://doi.org/10.1007/s12667-022-00513-8.

[146] G. Sharma, I. Nasiruddin, K.R. Niazi, R.C. Bansal. ANFIS based control design for AGC of a hydro-hydro power system with UPFC and hydrogen electrolyzer units. Electric Power Components and Systems. 46 (2018) 406-17. <u>https://doi.org/10.1080/15325008.2018.1446197</u>.

[147] S.M. Collura, D. Guilbert, G. Vitale, M. Luna, F. Alonge, F. d'Ippolito, A. Scipioni. Design and experimental validation of a high voltage ratio DC/DC converter for proton exchange membrane electrolyzer applications. International Journal of Hydrogen Energy. 44 (2019) 7059-72. <u>https://doi.org/10.1016/j.ijhydene.2019.01.210</u>.

[148] Y.N. Anagreh, A. Alnassan, A. Radaideh. High Performance MPPT Approach for Off-Line PV System Equipped With Storage Batteries and Electrolyzer. International Journal of Renewable Energy Development. 10 (2021). https://doi.org/10.14710/ijred.2021.34131.

[149] N. Ali, X. Shen, H. Armghan, Y. Du. Hierarchical control combined with higher order sliding mode control for integrating wind/tidal/battery/hydrogen powered DC offshore microgrid. Journal of Energy Storage. 82 (2024) 110521. https://doi.org/10.1016/j.est.2024.110521.

[150] B.T. Gul, S.H. Ahmed, I. Ahmad, O. Zeb. Optimized barrier-condition nonlinear control of wireless chargerbased hybrid electric vehicle. Journal of Energy Storage. 97 (2024) 112705. https://doi.org/10.1016/j.est.2024.112705.

[151] H. Armghan, M. Yang, N. Ali, A. Armghan, A. Alanazi. Quick reaching law based global terminal sliding mode control for wind/hydrogen/battery DC microgrid. Applied Energy. 316 (2022) 119050. https://doi.org/10.1016/j.apenergy.2022.119050.

[152] R. Mas, D. Pichilingue, A. Berastain, C. Celis. On the application of sliding mode control to indirectly coupled photovoltaic-electrolyzer systems used in the production of clean energy. International Journal of Thermofluids. 23 (2024) 100747. <u>https://doi.org/10.1016/j.ijft.2024.100747</u>.

[153] N. Chaitanya, S. Yamparala, K. Radharani, P.P. Prasanthi. Optimized power management control scheme for transportation system electrified with high voltage DC microgrid. International Journal of Renewable Energy Research. 12 (2022) 536-46. <u>https://doi.org/10.20508/ijrer.v12i1.12793.g8427</u>.

[154] A. Baraean, M. Kassas, M.S. Alam, M.A. Abido. Physics-informed NN-based adaptive backstepping terminal sliding mode control of buck converter for PEM electrolyzer. Heliyon. 10 (2024). https://doi.org/10.1016/j.heliyon.2024.e29254.

[155] K. Liaqat, Z. Rehman, I. Ahmad. Nonlinear controllers for fuel cell, photovoltaic cell and battery based hybrid energy management system. Journal of Energy Storage. 32 (2020) 101796. https://doi.org/10.1016/j.est.2020.101796.

[156] P. Yin, J. Guo, H. He. Backstepping sliding-mode techniques in current control of polymer electrolyte membrane fuel cell. Control Engineering Practice. 124 (2022) 105188. https://doi.org/10.1016/j.conengprac.2022.105188.

[157] Z. Zhu, X. Liu, X. Kong, L. Ma, K.Y. Lee, Y. Xu. PV/Hydrogen DC microgrid control using distributed economicmodelpredictivecontrol.RenewableEnergy.222(2024)119871.https://doi.org/10.1016/j.renene.2023.119871.

[158] L. Martínez, D. Fernández, R. Mantz. Two layer control strategy of an island DC microgrid with hydrogen storage system. International Journal of Hydrogen Energy. 50 (2024) 365-78. https://doi.org/10.1016/j.ijhydene.2023.09.009.

[159] R.R. Makineni, D. Sutanto, K.M. Muttaqi, M.R. Islam, A.P. Agalgaonkar. Dual loop cascade control of a
stacked interleaved buck converter for electrolyzer application. 2022 IEEE IAS Global Conference on Emerging
Technologies (GlobConET).IEEE2022.pp.1029-35.doi:https://doi.org/10.1109/GlobConET53749.2022.9872460.IEEE2022.IEEE2022.IEEE2022.IEEE2022.

[160] N. Dahmane, A. Djerioui, S. Zeghlache, Z.E. Dahmane, A. Houari, A. Hadjkaddour. Energy Managementof PV/FC/BAT/SC and PEM Electrolyzer. 2022 IEEE International Conference on Electrical Sciences andTechnologiesinMaghreb(CISTEM).IEEE2022.pp.1-6.doi:https://doi.org/10.1109/CISTEM55808.2022.10043977.

[161] B. Yodwong, P. Mungporn, S. Sikkabut, D. Guilbert, M. Phattanasak, M. Hinaje, G. Vitale, P. Thounthong.
 Performance Analysis of a Control Strategy for a Three-Level Interleaved Buck Converter for Proton Exchange
 Membrane Electrolyzer Applications. 2023 IEEE Transportation Electrification Conference and Expo, Asia Pacific (ITEC Asia-Pacific). IEEE2023. pp. 1-5. doi:<u>https://doi.org/10.1109/ITECAsia Pacific59272.2023.10372263</u>.

[162] M.B. Hossain, M.R. Islam, K.M. Muttaqi, D. Sutanto, A.P. Agalgaonkar. A power dispatch allocation strategy to produce green hydrogen in a grid-integrated offshore hybrid energy system. International Journal of Hydrogen Energy. 62 (2024) 1103-12. <u>https://doi.org/10.1016/j.ijhydene.2024.03.051</u>.

[163] Z. Zhu, S. Chen, X. Kong, L. Ma, X. Liu, K.Y. Lee. A centralized EMPC scheme for PV-powered alkaline electrolyzer. Renewable Energy. 229 (2024) 120688. <u>https://doi.org/10.1016/j.renene.2024.120688</u>.

[164] P. Rullo, L. Braccia, P. Luppi, D. Zumoffen, D. Feroldi. Integration of sizing and energy management based on economic predictive control for standalone hybrid renewable energy systems. Renewable energy. 140 (2019) 436-51. <u>https://doi.org/10.1016/j.renene.2019.03.074</u>.

[165] Q. Li, X. Zou, Y. Pu, W. Chen. Real-time Energy Management Method for Electric-hydrogen Hybrid Energy Storage Microgrids Based on DP-MPC. CSEE Journal of Power and Energy Systems. 10 (2020) 324-36. https://doi.org/10.17775/CSEEJPES.2020.02160.

[166] M.A. Syed, M. Khalid. A feedforward neural network hydrogen electrolyzer output regulator for wind power control with battery storage. 2021 IEEE PES Innovative Smart Grid Technologies-Asia (ISGT Asia). IEEE2021. pp. 1-5. doi:<u>https://doi.org/10.1109/ISGTAsia49270.2021.9715687</u>.

[167] B. Yodwong, D. Guilbert, W. Kaewmanee, M. Phattanasak, M. Hinaje, G. Vitale. Improved Sliding Mode-Based Controller of a High Voltage Ratio DC–DC Converter for Electrolyzers Supplied by Renewable Energy. IEEE Transactions on Industrial Electronics. (2023). https://doi.org/10.1109/TIE.2023.3322009.

[168] P. Bhat Nempu, N. Jayalakshmi. Coordinated power management of the subgrids in a hybrid AC–DC microgrid with multiple renewable sources. IETE Journal of Research. 68 (2022) 2790-800. https://doi.org/10.1080/03772063.2020.1726829.

[169] D.B. Aeggegn, G.N. Nyakoe, C. Wekesa. ANFIS-Controlled Boost and Bidirectional Buck-Boost DC-DC Converters for Solar PV, Fuel Cell, and BESS-Based Microgrid Application. International Transactions on Electrical Energy Systems. 2024 (2024) 6484369. <u>https://doi.org/10.1155/2024/6484369</u>.

[170] Y.R. Sankar, K. Chandra Sekhar. Adaptive Cascaded ANFIS MPPT Development for Solar and Fuel Cell Based Hybrid Energy System. Journal of The Institution of Engineers (India): Series B. (2024) 1-14. https://doi.org/10.1007/s40031-024-01090-3.

[171] S.A. Memon, R.N. Patel. An overview of optimization techniques used for sizing of hybrid renewable energy systems. Renewable Energy Focus. 39 (2021) 1-26. <u>https://doi.org/10.1016/j.ref.2021.07.007</u>.

[172] M.A. Nasab, M.A. Dashtaki, B. Ehsanmaleki, M. Zand, M.A. Nasab, P. Sanjeevikumar. LFC of smart, interconnected power system in the presence of renewable energy sources using coordinated control design of hybrid electric vehicles. Renewable Energy Focus. 50 (2024) 100609. https://doi.org/10.1016/j.ref.2024.100609.

[173] M. Ferrara, M. Bilardo, D.-I. Bogatu, D. Lee, M. Khatibi, S. Rahnama, J. Shinoda, Y. Sun, Y. Sun, A. Afshari. Review on Advanced Storage Control Applied to Optimized Operation of Energy Systems for Buildings and Districts: Insights and Perspectives. Energies. 17 (2024) 3371. <u>https://doi.org/10.3390/en17143371</u>.

[174] F.M. Almasoudi, A. Bakeer, G. Magdy, K.S.S. Alatawi, G. Shabib, A. Lakhouit, S.E. Alomrani. Nonlinear coordination strategy between renewable energy sources and fuel cells for frequency regulation of hybrid power systems. Ain Shams Engineering Journal. 15 (2024) 102399. https://doi.org/10.1016/j.asej.2023.102399.

[175] G. Sahoo, R.K. Sahu, S. Panda, N.R. Samal, Y. Arya. Modified Harris Hawks optimization-based fractionalorder fuzzy PID controller for frequency regulation of multi-micro-grid. Arabian Journal for Science and Engineering. 48 (2023) 14381-405. <u>https://doi.org/10.1007/s13369-023-07613-2</u>.

[176] S. Vishnoi, S. Nikolovski, M. Raju, M.K. Kirar, A.S. Rana, P. Kumar. Frequency stabilization in an interconnected micro-grid using smell agent optimization algorithm-tuned classical controllers considering electric vehicles and wind turbines. Energies. 16 (2023) 2913. <u>https://doi.org/10.3390/en16062913</u>.

[177] F. Alshehri, V.G. Suárez, J.L.R. Torres, A. Perilla, M. van der Meijden. Modelling and evaluation of PEM hydrogen technologies for frequency ancillary services in future multi-energy sustainable power systems. Heliyon. 5 (2019). <u>https://doi.org/10.1016/j.heliyon.2019.e01396</u>.

[178] A. Saha, M.S. Bhaskar, D.J. Almakhles, M.F. Elmorshedy. Employment of renewable based sources in amalgamated frequency-voltage control restructured system with TSA trained IPD (1+ I) controller. Renewable Energy. 222 (2024) 119879. https://doi.org/10.1016/j.renene.2023.119879.

[179] S. Mishra, P.C. Nayak, R.C. Prusty, S. Panda. Design and Analysis of Time-Varying Derivative Fractional Order Proportional–Integral–Derivative Controller for Frequency Regulation of Shipboard Microgrid System. Electrica. 24 (2024). https://doi.org/10.5152/electrica.2024.23055.

[180] G.-H. Lee, J.-Y. Park, J. Ban, Y.-J. Kim, J.P. Catalão. Data-driven modeling and optimal control of hydrogen energy storage for frequency regulation. IEEE Transactions on Energy Conversion. 38 (2022) 1231-45. https://doi.org/10.1109/TEC.2022.3221165.

[181] S. Ghosh, R. Jain, S. Veda, D. Terlip. Improving distribution system operations using fleet control of electrolyzers. 2020 IEEE/PES Transmission and Distribution Conference and Exposition (T&D). IEEE2020. pp. 1-5. doi:<u>https://doi.org/10.1109/TD39804.2020.9300018</u>.

[182] W. Zhao, M.R. Nielsen, M. Kjær, F. Iov, S.M. Nielsen. Grid integration of a 500 kW alkaline electrolyzer system for harmonic analysis and robust control. e-Prime-Advances in Electrical Engineering, Electronics and Energy. 5 (2023) 100217. <u>https://doi.org/10.1016/j.prime.2023.100217</u>.

[183] R. Jakobsen, C. Huang, T.W. Rasmussen, S. You. Large-scale electrolyzer plant integration to the electrical grid: Preliminary investigation of VSC-based solutions. Energy Reports. 9 (2023) 478-83. https://doi.org/10.1016/j.egyr.2023.09.061.

[184] R. Loka, A.M. Parimi, S. Srinivas, N.M. Kumar. Leveraging blockchain technology for resilient and robust frequency control in a renewable-based hybrid power system with hydrogen and battery storage integration. Energy Conversion and Management. 283 (2023) 116888. <u>https://doi.org/10.1016/j.enconman.2023.116888</u>.
[185] H.S. Salama, G. Magdy, A. Bakeer, I. Vokony. Adaptive coordination control strategy of renewable energy sources, hydrogen production unit, and fuel cell for frequency regulation of a hybrid distributed power system. Protection and Control of Modern Power Systems. 7 (2022) 1-18. <u>https://doi.org/10.1186/s41601-022-00258-7</u>.

[186] F. Zenith, M.N. Flote, M. Santos-Mugica, C.S. Duncan, V. Mariani, C. Marcantonini. Value of green hydrogen when curtailed to provide grid balancing services. international journal of hydrogen energy. 47 (2022) 35541-52. <u>https://doi.org/10.1016/j.ijhydene.2022.08.152</u>.

[187] H. Hobbie, M. Lieberwirth. Compounding or Curative? Investigating the impact of electrolyzer deployment on congestion management in the German power grid. Energy Policy. 185 (2024) 113900. https://doi.org/10.1016/j.enpol.2023.113900.

[188] S. Ghaemi, X. Li, M. Mulder. Economic feasibility of green hydrogen in providing flexibility to medium-voltage distribution grids in the presence of local-heat systems. Applied Energy. 331 (2023) 120408. https://doi.org/10.1016/j.apenergy.2022.120408.

[189] M. Lieberwirth, H. Hobbie. Decarbonizing the industry sector and its effect on electricity transmission grid operation—Implications from a model based analysis for Germany. Journal of Cleaner Production. 402 (2023) 136757. <u>https://doi.org/10.1016/j.jclepro.2023.136757</u>.

[190] A. Heidari-Akhijahani, K.L. Butler-Purry. A review on black-start service restoration of active distribution systems and microgrids. Energies. 17 (2023) 100. <u>https://doi.org/10.3390/en17010100</u>.

[191] J. Fan, L. Niu, C. Li, G. Zhang, H. Li, Y. Wang, J. Li, Q. Song, J. Sun, J. Pan. Review of Black Start on New Power System Based on Energy Storage Technology. Energy Engineering. 120 (2023). https://doi.org/10.32604/ee.2023.029740.

[192] C.-M. Lai, J. Teh, B. Alharbi, A. AlKassem, A. Aljabr, N. Alshammari. Optimisation of generation unit commitment and network topology with the dynamic thermal rating system considering N-1 reliability. Electric Power Systems Research. 221 (2023) 109444.

[193] C.-M. Lai, J. Teh. Network topology optimisation based on dynamic thermal rating and battery storage systems for improved wind penetration and reliability. Applied Energy. 305 (2022) 117837.

[194] L. Yang, J. Teh, B. Alharbi. Optimizing distributed generation and energy storage in distribution networks: Harnessing metaheuristic algorithms with dynamic thermal rating technology. Journal of Energy Storage. 91 (2024) 111989.

[195] Y. Su, J. Teh, Q. Luo, K. Tan, J. Yong. A two-layer framework for mitigating the congestion of urban power grids based on flexible topology with dynamic thermal rating. Protection and Control of Modern Power Systems. (2024).

[196] Y. Su, J. Teh, C. Chen. Optimal dispatching for AC/DC hybrid distribution systems with electric vehicles: application of cloud-edge-device cooperation. IEEE Transactions on Intelligent Transportation Systems. (2023).

[197] Y. Su, J. Teh, W. Liu. Hierarchical and distributed energy management framework for AC/DC hybrid distribution systems with massive dispatchable resources. Electric Power Systems Research. 225 (2023) 109856.

[198] S. Mishra, P.C. Nayak, R.C. Prusty, S. Panda. Modified multiverse optimizer technique-based two degree of freedom fuzzy PID controller for frequency control of microgrid systems with hydrogen aqua electrolyzer fuel cell unit. Neural Computing and Applications. 34 (2022) 18805-21. <u>https://doi.org/10.1007/s00521-022-07453-5</u>.

[199] S. Yıldız, H. Gunduz, B. Yildirim, M.T. Özdemir. An islanded microgrid energy system with an innovative frequency controller integrating hydrogen-fuel cell. Fuel. 326 (2022) 125005. https://doi.org/10.1016/j.fuel.2022.125005.

[200] P.C. Sahu, R.C. Prusty, S. Panda. Improved-GWO designed FO based type-II fuzzy controller for frequency awareness of an AC microgrid under plug in electric vehicle. Journal of Ambient Intelligence and Humanized Computing. 12 (2021) 1879-96. <u>https://doi.org/10.1007/s12652-020-02260-z</u>.

[201] B. Dhanasekaran, S. Siddhan, J. Kaliannan. Ant colony optimization technique tuned controller for frequency regulation of single area nuclear power generating system. Microprocessors and Microsystems. 73 (2020) 102953. https://doi.org/10.1016/j.micpro.2019.102953.

[202] M. Maluenda, S. Córdova, Á. Lorca, M. Negrete-Pincetic. Optimal operation scheduling of a PV-BESS-Electrolyzer system for hydrogen production and frequency regulation. Applied Energy. 344 (2023) 121243. https://doi.org/10.1016/j.apenergy.2023.121243.

[203] Z. Li, Z. Tu, Z. Yi, Y. Xu. Coordinated Control of Proton Exchange Membrane Electrolyzers and Alkaline Electrolyzers for a Wind-to-Hydrogen Islanded Microgrid. Energies. 17 (2024) 2317. https://doi.org/10.3390/en17102317.

[204] M.A. Syed, M. Khalid. An Intelligent Model Predictive Control Strategy for Stable Solar-Wind Renewable Power Dispatch Coupled with Hydrogen Electrolyzer and Battery Energy Storage. International Journal of Energy Research. 2023 (2023) 4531054. <u>https://doi.org/10.1155/2023/4531054</u>.

[205] C. Huang, Y. Zong, S. You, C. Træholt. Analytical modeling and control of grid-scale alkaline electrolyzer plant for frequency support in wind-dominated electricity-hydrogen systems. IEEE Transactions on Sustainable Energy. 14 (2022) 217-32. <u>https://doi.org/10.1109/TSTE.2022.3208361</u>.

[206] M. Agredano-Torres, M. Zhang, L. Söder, Q. Xu. Decentralized Dynamic Power Sharing Control for Frequency Regulation Using Hybrid Hydrogen Electrolyzer Systems. IEEE Transactions on Sustainable Energy. (2024). <u>https://doi.org/10.1109/TSTE.2024.3381491</u>.

[207] M.B. Hossain, M.R. Islam, K.M. Muttaqi, D. Sutanto, A.P. Agalgaonkar. A compensation strategy for mitigating intermittencies within a PV powered microgrid using a hybrid multilevel energy storage system. IEEE Transactions on Industry Applications. 59 (2023) 5074-86. <u>https://doi.org/10.1109/TIA.2023.3262771</u>.

[208] F. Tong, W. Wei, M. Jin, X. Sun, Q. Wang, Q. Luo. Optimal Fractional-Order Proportion Integration Differentiation Control of Proton Exchange Membrane Electrolyzer for Offshore Wind Power Hydrogen Production System. Energy Technology. 11 (2023) 2300100. <u>https://doi.org/10.1002/ente.202300100</u>.

[209] M.K. Debnath, R. Agrawal, S.R. Tripathy, S. Choudhury. Artificial neural network tuned PID controller for LFC investigation including distributed generation. International Journal of Numerical Modelling: Electronic Networks, Devices and Fields. 33 (2020) e2740. <u>https://doi.org/10.1002/jnm.2740</u>.

[210] M.S. Nazir, A.N. Abdalla, H. Zhao, Z. Chu, H.M.J. Nazir, M.S. Bhutta, M.S. Javed, P. Sanjeevikumar. Optimized economic operation of energy storage integration using improved gravitational search algorithm and dual stage optimization. Journal of Energy Storage. 50 (2022) 104591.

[211] F. Gutiérrez-Martín, L. Amodio, M. Pagano. Hydrogen production by water electrolysis and off-grid solar PV. International Journal of Hydrogen Energy. 46 (2021) 29038-48. https://doi.org/10.1016/j.ijhydene.2020.09.098.

[212] A. Grimm, W.A. de Jong, G.J. Kramer. Renewable hydrogen production: A techno-economic comparison of photoelectrochemical cells and photovoltaic-electrolysis. International Journal of Hydrogen Energy. 45 (2020) 22545-55. <u>https://doi.org/10.1016/j.ijhydene.2020.06.092</u>.

[213] H. Niaz, M.M. Lakouraj, J. Liu. Techno-economic feasibility evaluation of a standalone solar-powered alkaline water electrolyzer considering the influence of battery energy storage system: A Korean case study. Korean Journal of Chemical Engineering. 38 (2021) 1617-30. https://doi.org/10.1007/s11814-021-0819-z.

[214] C. Schnuelle, T. Wassermann, D. Fuhrlaender, E. Zondervan. Dynamic hydrogen production from PV & wind direct electricity supply–Modeling and techno-economic assessment. International Journal of Hydrogen Energy. 45 (2020) 29938-52. https://doi.org/10.1016/j.ijhydene.2020.08.044.

[215] G. Yang, Y. Jiang, S. You. Planning and operation of a hydrogen supply chain network based on the offgrid wind-hydrogen coupling system. International Journal of Hydrogen Energy. 45 (2020) 20721-39. https://doi.org/10.1016/j.ijhydene.2020.05.207.

[216] S. Şevik. Techno-economic evaluation of a grid-connected PV-trigeneration-hydrogen production hybrid system on a university campus. International Journal of Hydrogen Energy. 47 (2022) 23935-56. https://doi.org/10.1016/j.ijhydene.2022.05.193.

[217] H. Tebibel. Methodology for multi-objective optimization of wind turbine/battery/electrolyzer system for decentralized clean hydrogen production using an adapted power management strategy for low wind speed conditions. Energy Conversion and Management. 238 (2021) 114125. https://doi.org/10.1016/j.enconman.2021.114125.

[218] Z. Abdin, W. Mérida. Hybrid energy systems for off-grid power supply and hydrogen production based on renewable energy: A techno-economic analysis. Energy Conversion and management. 196 (2019) 1068-79. https://doi.org/10.1016/j.enconman.2019.06.068.

[219] M. Temiz, I. Dincer. Techno-economic analysis of green hydrogen ferries with a floating photovoltaic based marine fueling station. Energy Conversion and Management. 247 (2021) 114760. https://doi.org/10.1016/j.enconman.2021.114760.

[220] P. Benalcazar, A. Komorowska. Prospects of green hydrogen in Poland: A techno-economic analysis using a Monte Carlo approach. International Journal of Hydrogen Energy. 47 (2022) 5779-96. https://doi.org/10.1016/j.ijhydene.2021.12.001.

[221] K. Almutairi, S.S.H. Dehshiri, S.J.H. Dehshiri, A. Mostafaeipour, M. Jahangiri, K. Techato. Technical, economic, carbon footprint assessment, and prioritizing stations for hydrogen production using wind energy: A case study. Energy Strategy Reviews. 36 (2021) 100684. <u>https://doi.org/10.1016/j.esr.2021.100684</u>.

[222] M. Nasser, T.F. Megahed, S. Ookawara, H. Hassan. Techno-economic assessment of clean hydrogen production and storage using hybrid renewable energy system of PV/Wind under different climatic conditions. Sustainable Energy Technologies and Assessments. 52 (2022) 102195. https://doi.org/10.1016/j.seta.2022.102195.

[223] J. Šimunović, I. Pivac, F. Barbir. Techno-economic assessment of hydrogen refueling station: A case study in Croatia. International journal of hydrogen energy. 47 (2022) 24155-68. https://doi.org/10.1016/j.ijhydene.2022.05.278.

[224] R. Babaei, D.S. Ting, R. Carriveau. Optimization of hydrogen-producing sustainable island microgrids.InternationalJournalofHydrogenEnergy.47(2022)14375-92.https://doi.org/10.1016/j.ijhydene.2022.02.187.

[225] S. Praveenkumar, E.B. Agyekum, J.D. Ampah, S. Afrane, V.I. Velkin, U. Mehmood, A.A. Awosusi. Technoeconomic optimization of PV system for hydrogen production and electric vehicle charging stations under five different climatic conditions in India. International Journal of Hydrogen Energy. 47 (2022) 38087-105. https://doi.org/10.1016/j.ijhydene.2022.09.015.

[226] J. Park, S. Kang, S. Kim, H.-S. Cho, S. Heo, J.H. Lee. Techno-economic analysis of solar powered green hydrogen system based on multi-objective optimization of economics and productivity. Energy Conversion and Management. 299 (2024) 117823. <u>https://doi.org/10.1016/j.enconman.2023.117823</u>.

[227] R. Boudries. Techno-economic study of hydrogen production using CSP technology. International Journal of Hydrogen Energy. 43 (2018) 3406-17. <u>https://doi.org/10.1016/j.ijhydene.2017.05.157</u>.

[228] R. Bhandari, R.R. Shah. Hydrogen as energy carrier: Techno-economic assessment of decentralized hydrogen production in Germany. Renewable Energy. 177 (2021) 915-31. https://doi.org/10.1016/j.renene.2021.05.149. [229] M.S. Salhi, P.C. Okonkwo, I.B. Belgacem, S. Farhani, M. Zghaibeh, F. Bacha. Techno-economic optimization of wind energy based hydrogen refueling station case study Salalah city Oman. International Journal of Hydrogen Energy. 48 (2023) 9529-39. <u>https://doi.org/10.1016/j.ijhydene.2022.12.148</u>.

[230] E. Assareh, A. Ghafouri. An innovative compressed air energy storage (CAES) using hydrogen energy integrated with geothermal and solar energy technologies: a comprehensive techno-economic analysis-different climate areas-using artificial intelligent (AI). International Journal of Hydrogen Energy. 48 (2023) 12600-21. https://doi.org/10.1016/j.ijhydene.2022.11.233.

[231] A. Kim, H. Kim, C. Choe, H. Lim. Feasibility of offshore wind turbines for linkage with onshore green hydrogen demands: A comparative economic analysis. Energy Conversion and Management. 277 (2023) 116662. <u>https://doi.org/10.1016/j.enconman.2023.116662</u>.

[232] M. Nasser, T.F. Megahed, S. Ookawara, H. Hassan. Performance evaluation of PV panels/wind turbines hybrid system for green hydrogen generation and storage: Energy, exergy, economic, and enviroeconomic. Energy Conversion and Management. 267 (2022) 115870. <u>https://doi.org/10.1016/j.enconman.2022.115870</u>.
[233] G. Mensah, R. Opoku, F. Davis, G.Y. Obeng. Techno-economic analysis of green hydrogen production and electric vehicle charging using redundant energy on a solar photovoltaic mini-grid. Cleaner Energy Systems. 9 (2024) 100165. <u>https://doi.org/10.1016/j.cles.2024.100165</u>.

[234] M. Al-Mahmodi, O. Ayadi, Y. Wang. Sensitivity-based techno-economic assessment approach for electrolyzer integration with hybrid photovoltaic-wind plants for green hydrogen production. International Journal of Hydrogen Energy. 97 (2025) 904-19. <u>https://doi.org/10.1016/j.ijhydene.2024.12.002</u>.